World health statistics 2025

Monitoring health for the SDGs, Sustainable Development Goals





World health statistics 2025

Monitoring health for the SDGs, Sustainable Development Goals



World health statistics 2025: monitoring health for the SDGs, Sustainable Development Goals

ISBN 978-92-4-011049-6 (electronic version) ISBN 978-92-4-011050-2 (print version)

© World Health Organization 2025

Some rights reserved. This work is available under the Creative Commons Attribution-NonCommercial-ShareAlike 3.0 IGO licence (CC BY-NC-SA 3.0 IGO; https://creativecommons.org/licenses/by-nc-sa/3.0/igo).

Under the terms of this licence, you may copy, redistribute and adapt the work for non-commercial purposes, provided the work is appropriately cited, as indicated below. In any use of this work, there should be no suggestion that WHO endorses any specific organization, products or services. The use of the WHO logo is not permitted. If you adapt the work, then you must license your work under the same or equivalent Creative Commons licence. If you create a translation of this work, you should add the following disclaimer along with the suggested citation: "This translation was not created by the World Health Organization (WHO). WHO is not responsible for the content or accuracy of this translation. The original English edition shall be the binding and authentic edition".

Any mediation relating to disputes arising under the licence shall be conducted in accordance with the mediation rules of the World Intellectual Property Organization (http://www.wipo.int/amc/en/mediation/rules/).

Suggested citation. World health statistics 2025: monitoring health for the SDGs, Sustainable Development Goals. Geneva: World Health Organization; 2025. Licence: CC BY-NC-SA 3.0 IGO.

Cataloguing-in-Publication (CIP) data. CIP data are available at https://iris.who.int/.

Sales, rights and licensing. To purchase WHO publications, see https://www.who.int/publications/book-orders. To submit requests for commercial use and queries on rights and licensing, see https://www.who.int/copyright.

Third-party materials. If you wish to reuse material from this work that is attributed to a third party, such as tables, figures or images, it is your responsibility to determine whether permission is needed for that reuse and to obtain permission from the copyright holder. The risk of claims resulting from infringement of any third-party-owned component in the work rests solely with the user.

General disclaimers. The designations employed and the presentation of the material in this publication do not imply the expression of any opinion whatsoever on the part of WHO concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries. Dotted and dashed lines on maps represent approximate border lines for which there may not yet be full agreement.

The mention of specific companies or of certain manufacturers' products does not imply that they are endorsed or recommended by WHO in preference to others of a similar nature that are not mentioned. Errors and omissions excepted, the names of proprietary products are distinguished by initial capital letters.

All reasonable precautions have been taken by WHO to verify the information contained in this publication. However, the published material is being distributed without warranty of any kind, either expressed or implied. The responsibility for the interpretation and use of the material lies with the reader. In no event shall WHO be liable for damages arising from its use.

Design and layout: Green Ink Publishing Services Ltd.

Contents

Foreword			v			
Ack	Acknowledgements			vi		
Abl	Abbreviations Introduction					
Int						
Key messages						
1.	Chai	nge and	inequality in healthy longevity, and the contributing causes	5		
	1.1	Chang	je in HALE between 2000 and 2019	7		
		1.1.1	Global	7		
		1.1.2	WHO regions	8		
	1.2	Chang	e in HALE during the COVID-19 pandemic in 2019–2021	10		
		1.2.1	Change attributed to mortality	10		
		1.2.2	Change attributed to morbidity	11		
	1.3	Gap in	HALE between males and females in 2019 and 2021	12		
		1.3.1	Global	12		
		1.3.2	WHO regions	14		
	1.4	Gap in	HALE between high-income and other income groups in 2019 and 2021	15		
	1.5	Progre	ess in premature mortality reduction	17		
		1.5.1	The prospect of premature mortality by 2050	18		
	Refe	rences		20		
2.	Heal	lth-relat	ed Sustainable Development Goals	21		
	2.1	Mortality-related Sustainable Development Goal indicators		22		
		2.1.1	Maternal and child mortality	22		
		2.1.2	Mortality due to noncommunicable diseases	24		
		2.1.3	Mortality due to injuries	25		
		2.1.4	Mortality attributable to environmental risk factors	27		
	2.2	Infecti	ous diseases	29		
		2.2.1	HIV	29		
		2.2.2	Tuberculosis	30		
		2.2.3	Malaria	31		
		2.2.4	Hepatitis B	31		
		2.2.5	Neglected tropical diseases	32		
		2.2.6	Antimicrobial resistance	33		
	2.3	Risk fa	ictors for health	34		
		2.3.1	Nutritional risk factors	34		
		2.3.2	Behavioural risk factors	36		
		2.3.3	Environmental risk factors	38		
		2.3.4	Risks to girls' and women's health	40		

	2.4	Universal health coverage and health systems		41
		2.4.1 Service delivery		41
		2.4.2 Health financing		45
	2.5	SDG progress to date		47
	Refe	rences		48
3.	Progress in achieving the Triple Billion targets			51
	3.1	Triple Billion targets progress		52
	3.2	Progress in healthier populations		53
	3.3	Progress in universal health coverage		54
	3.4	Progress in health emergencies protection		54
	3.5	Forecast to 2030 and scenarios		55
	3.6	Conclusion		57
	Refe	rences		58
4.	Inequality in immunization			59
	4.1	The importance of vaccine equity		60
	4.2	Inequalities in immunization coverage persist withi	n	
		countries		60
	4.3	How inequalities have changed over time		65
	4.4	Barriers to immunization		66
	4.5	Data availability to monitor inequalities in immuniz	ation	67
	Refe	rences		68
Ann	exes			69
Annex 1.		Progress assessment of selected health-related SDC	indicators by WHO region	70
Annex 2.		Summary of methodology		75

Foreword

To make progress, we must be able to measure progress. Data about the health of populations are foundational to understanding trends, making policies and directing resources to where they will have the greatest benefit. Without timely, trusted and actionable data, threats remain invisible, systems underperform and opportunities to save lives are lost.

World health statistics is the world's annual health report card. It shows where progress has been made – and where it has stalled. This year's edition highlights the impacts of the COVID-19 pandemic, which reversed many of the gains made in the previous two decades. Between 2000 and 2019, healthy life expectancy rose by more than five years, maternal mortality fell by one third, child mortality more than halved and premature deaths fell – driven by political commitment, investment, innovation and stronger health systems. In 2020 and 2021, COVID-19 killed millions of people, put health systems under severe strain and wiped 1.8 years off healthy life expectancy.

This report also shows that at the current pace, the world will miss the target in the Sustainable Development Goals (SDGs) to reduce premature deaths from noncommunicable diseases (NCDs) by one third. While mortality rates have declined, the number of premature NCD deaths continues to rise, due to population growth and ageing. Tobacco use is declining but not fast enough, while alcohol consumption has decreased in some regions but stagnated in others. Hypertension and diabetes remain inadequately controlled, and air pollution continues to claim millions of lives globally.

Likewise, progress on maternal and child mortality has slowed considerably, and many countries are off- course for the 2030 targets, due to underfunding of primary health care and huge gaps in access to essential services, such as skilled care at birth, immunization and access to health workers. Domestic investment must increase, particularly because where the need is greatest, resources are most limited. WHO's Thirteenth General Programme of Work translated the health-related targets in the SDGs into the "triple billion" targets for 2018–2025: one billion more people living healthier lives; one billion more people covered by universal health coverage without financial hardship; and one billion more people better protected from health emergencies. By the end of 2024, we estimate that the first target was already exceeded, but the second two are likely to be missed: about 1.4 billion more people were living healthier lives, but only 431 million more people were enjoying access to universal health coverage, and 637 million more people were better protected from health emergencies – encouraging progress, but less than is needed to reach the SDGs.

Nevertheless, it's not 2030 yet. With stronger leadership and scaled-up delivery, the gap can still be closed. Accordingly, the Fourteenth General Programme of Work sets ambitious new targets: to keep 6 billion people healthier, expand access to affordable health services for 5 billion people, and to see 7 billion people better protected from health emergencies between 2025 and 2028.

Health data are central to achieving these targets, which is why WHO is working with countries and partners to modernize information systems and link data to delivery, through the World Health Data Hub, the SCORE package of health data tools, and the Delivery for Impact approach.

Because every life counts.



1 ende

Dr Tedros Adhanom Ghebreyesus Director-General World Health Organization

Acknowledgements

This publication was produced by the WHO Division of Data, Analytics and Delivery for Impact in collaboration with WHO technical programmes and regional offices. WHO is grateful to UNICEF, the Population Division in the Department of Economic and Social Affairs of the United Nations and the Joint United Nations Programme on HIV/AIDS (UNAIDS) for their contributions.

Abbreviations

ABR	adolescent birth rate	MNT	maternal and neonatal tetanus
AMR	antimicrobial resistance	MRSA	methicillin-resistant Staphylococcus aureus
APC	alcohol per capita consumption	NCDs	noncommunicable diseases
ARIs	acute respiratory infections	NMR	neonatal mortality rate
ARR	annual rate of reduction	NTDs	neglected tropical diseases
ART	antiretroviral therapy	ODA	official development assistance
ASR	age-standardized death rate	OOP	out-of-pocket (health spending)
CDR	crude death rate	PAB	protection at birth
COVID-19	coronavirus disease	PCV3	pneumococcal-conjugate vaccine third dose
DHS	Demographic and Health Survey(s)	PM2.5	fine particulate matter less than 2.5 µm
DTP3	diphtheria-tetanus-pertussis vaccine		in diameter
CLASS	third dose	RMNCH	reproductive, maternal, newborn and
GLASS	Surveillance System	SDC.	Sustainable Development Cool
GPW13	Thirteenth General Programme of Work	SDG	Sustainable Development Goal
GPW14	Fourteenth General Programme of Work	SPAR	Reporting Tool
GTS	Global technical strategy for	SPL	societal poverty line
	malaria 2016–2030	тв	tuberculosis
HALE	healthy life expectancy	U5MR	under-five mortality rate
HIC	high-income country	UHC	universal health coverage
IA2030	Immunization Agenda 2030	UI	uncertainty interval
IHRs	international health regulations	UMIC	upper-middle-income country
IQR	interquartile range	UNAIDS	Joint United Nations Programme
LIC	low-income country		on HIV/AIDS
LMIC	lower-middle-income country	USAID	United States Agency for
MCV2	measles-containing vaccine second dose		International Development
MIC	middle-income country	WASH	water, sanitation and hygiene
MICS	Multiple Indicator Cluster Survey(s)	WHA	World Health Assembly
MMR	maternal mortality ratio	WHO	World Health Organization

Introduction

The *World health statistics* report is the annual compilation of health and health-related indicators, which has been published by the World Health Organization (WHO) since 2005.

The 2025 edition consists of four chapters and the accompanying annexes. Chapter 1 presents an indepth analysis of global and regional estimates of life expectancy, healthy life expectancy and progress in the reduction of premature mortality. Chapter 2 reviews the status of the health-related Sustainable Development Goal (SDG) indicators, covering age- and cause-specific mortality, infectious diseases, risk factors for health, and universal health coverage (UHC) and health systems. A summary of global and regional trends towards the achievement of selected health-related SDG indicators is further presented in Annex 1. Chapter 3 presents updated projections on WHO's Triple Billion targets, based on the most recent available data. It also describes WHO's transition from its thirteenth general programme of work (GPW13) to the fourteenth general programme of work (GPW14), aligning with emerging global health challenges and priorities. Chapter 4 looks at inequalities in immunization as an important global health issue.

The information presented in *World health statistics 2025* is based on data available from global monitoring as of April 2025. These data have been compiled primarily from publications and databases produced and maintained by WHO, other United Nations entities (such as UNICEF, UNAIDS, and the Population Division in the Department of Economic and Social Affairs of the United Nations), United Nations Inter-Agency bodies of which WHO is a member, and other international organizations. A summary of the methodology for each chapter is described in Annex 2. Tables of health statistics by country and area, WHO region and globally can be accessed via https://www.who.int/data/gho/publications/ world-health-statistics.

Key messages

The major contributors to the 5.4-year pre-pandemic increase in global healthy life expectancy (HALE) at birth between 2000 (58.1 years) and 2019 (63.5 years) were mortality reduction from communicable and perinatal conditions among ages under 5 years, and from noncommunicable diseases (NCDs) among those 30 years and older. However, worsening morbidity due to diabetes among ages 30 years and older has led to a 0.14-year loss in HALE. Each WHO region exhibits distinctive patterns given their unique compositions of mortality and morbidity.

Between 2019 and 2021, the HALE loss can be almost entirely explained by mortality both directly and indirectly attributable to coronavirus disease (COVID-19) among those aged 30 years and older, both globally and in most WHO regions. The pandemic also caused increased morbidity from anxiety disorders and depressive disorders, each responsible for a 0.06-year (or 3-week) loss in global HALE in 2019–2021, and together effectively wiped out 80% of the positive contribution to HALE (0.15 years, or 8 weeks) by mortality decline from all NCDs combined in that two-year period.

The contributors to the HALE difference between males and females show a mixed pattern. While lower female mortality from injuries among 5–69-year-olds and from NCDs including ischaemic heart disease and stroke among females of 30 years and older gives a lead in HALE compared with males, mortality in maternal conditions and breast cancers, and higher female morbidity from conditions including back and neck pain, gynaecological diseases, migraine, depressive disorders and anxiety disorders have effectively eliminated a considerable share of the female HALE lead.

While lower mortality from communicable and perinatal causes explains a significant amount of the HALE lead that populations in high-income countries (HICs) had against those in low-income countries (LICs) and lower-middle-income countries (LMICs), higher mortality from tracheal, bronchus and lung cancers and drug use disorders, and higher morbidity from falls and back and neck pain has cost HICs some loss in HALE compared with other income groups.

Premature mortality was in decline globally and in all WHO regions from 2000 to the onset of the COVID-19 pandemic. Despite the overall progress prior to the pandemic, stagnation was seen after 2015. Globally, the pace of decline is projected to slow between now and 2050, compared with the rates observed in 2000–2019. Only about one third of countries are projected to have accelerated progress in 2019–2050 compared with 2000–2019, mostly located in the Region of the Americas and the Western Pacific Region.

As the world moves closer to 2030, overall progress is insufficient to meet the health-related SDG and other global targets. Declines were observed in mortality from causes addressed by the SDG indicators, including maternal mortality, child and neonatal mortality, premature NCD mortality, injury mortality, and mortality attributed to unsafe water, sanitation and hygiene (WASH) and air pollution. However, the progress for all these indicators is either insufficient or stalled, and is currently off-track for achieving their respective global targets.

There has been mixed progress in the global fight against infectious diseases, with declines in HIV and tuberculosis (TB) incidence rates, as well as in the number of people requiring interventions against neglected tropical diseases (NTDs), while malaria incidence rates have increased since 2015 and hepatitis B and antimicrobial resistance (AMR) continue to pose challenges.

Similarly, the world is off-track for meeting most of the global targets in reducing the prevalence or exposure to health risk factors such as malnutrition, tobacco use, unsafe WASH, air pollution and violence against women and girls, despite improvement in some of these areas.

However, the world is on-track to reach the 20% reduction in total per-capita alcohol consumption by 2030, although progress differs by region.

While gains in UHC service coverage index (SDG indicator 3.8.1) were observed globally over the past two decades, progress has slowed in recent years. In 2019, some 344 million people were pushed or further pushed into extreme poverty by out-of-pocket (OOP) health spending, while 13.5% of the global population spent more than 10% of their household budget on OOP payments for health.

The estimated global shortage of health workers of 15.4 million in 2020 decreased to 14.7 million in 2023. The projected 2030 shortage of 11.1 million shows slow progress in closing the gap, with the WHO African and Eastern Mediterranean regions projected to bear nearly 70% of the shortage.

Progress towards achieving the Triple Billion targets remains uneven. The expansion of UHC is insufficient, with only about 431 million additional people gaining access to essential health services without incurring financial hardship by 2024 compared with the 2018 baseline set in the WHO Thirteenth General Programme of Work, and only a projected 500 million by 2025 – just half of the targeted one billion. Meanwhile, close to 637 million additional individuals are expected to be better protected from health emergencies by 2024. This is expected to increase to only 697 million in 2025 - a substantial gain that nonetheless remains a little over 30% short of the one-billion target. On the plus side, an estimated 1.35 billion more people will experience healthier lives by 2024, rising to 1.5 billion by 2025, thus exceeding the original target of one billion. However, this progress remains insufficient to put the world on track to achieve the health-related SDGs by 2030. Moreover, recent interruptions in international aid threaten to disrupt health services and systems, disproportionately impacting countries and communities with the greatest health-care needs. Safeguarding the gains made

towards the Triple Billion targets remains of paramount importance for the global community in the years ahead.

Within-country inequalities in childhood immunization related to economic status and education level of the mother persist. Positively, inequalities in the last decade have diminished, compared with those in the previous decade, in LICs and LMICs. For instance, economic-related inequality in DTP3 coverage and zero-dose prevalence have reduced by more than half across LICs. Eliminating economic-related inequality across 88 LICs and middleincome countries is associated with improving the national average DTP3 coverage across these countries by 10 percentage points and halving zero-dose prevalence. Achieving equity in immunization coverage requires focused attention on urban poor areas, remote/ rural areas, conflict areas, and gender-related inequities and barriers.

Change and inequality in healthy longevity, and the contributing causes

Steady increases in life expectancy and healthy life expectancy (HALE) at birth were observed globally in 2000–2019, followed by rapid declines between 2019 and 2021 due to the COVID-19 pandemic. Various factors exhibit diverse epidemiological patterns across different stages in an individual's life course and in different populations, contributing in various degrees to the change in life expectancy and HALE over time or the inequality therein between populations. This chapter reviews the contribution of different causes to the change in HALE before and during the COVID-19 pandemic, and to the inequality in HALE between males and females and between income groups as defined by the World Bank. Unless otherwise stated, the principal source of data for this chapter is the World Health Organization (WHO) Global health estimates 2021 (1).

The global community enjoyed steady improvements in many aspects of population health between the turn of the century until the onset of the pandemic, from increasing access to clean water and sanitation to declining prevalence of tobacco smoking, from rising proportion of births attended by skilled health personnel (2) to doubling effective coverage for hypertension treatment (3), and from dropping mortality due to HIV/AIDS to improving child survival (4,5). These improvements led to a decline in mortality from a broad spectrum of causes of death throughout the human lifespan.

Both life expectancy and HALE increased between 2000 and 2019. Global life expectancy at birth increased by 6.3 years in this period, from 66.8 years in 2000 to 73.1 years in 2019, with males gaining 6.2 years (from 64.4 to 70.6 years) and females gaining 6.5 years (from 69.2 to 75.7 years) during this period. Simultaneously, global HALE at birth increased by 5.4 years, from 58.1 years in 2000 to 63.5 years in 2019 (from 57.0 to 62.3 years for males and from 59.3 years to 64.6 years females) (Fig. 1.1).

However, this progress faced a major setback as the COVID-19 pandemic wreaked havoc across the globe. With COVID-19 becoming one of the leading causes of deaths globally and in many of the WHO regions in 2020 and 2021 and claiming an excess toll of lives indirectly through other causes, both life expectancy and HALE rolled back by nearly a decade within just two years. Global life expectancy at birth dropped by 0.7 years to 72.5 years in 2020 (back to the level of 2016), and by a further 1.1 years to 71.4 years in 2021 (back to the level of 2012). Similarly, global HALE at birth dropped to 62.8 years in 2020 (back to the level of 2016) and 61.9 years in 2021 (back to the level of 2012) (Fig. 1.1).

This chapter reviews the contribution of different causes to the change in HALE before and during the COVID-19 pandemic, and to the inequality in HALE between males and females and between income groups (6,7) as defined by the World Bank.



Figure 1.1 Life expectancy (LE) and healthy life expectancy at birth (HALE), by WHO region, 2000, 2019 and 2021

Source: WHO (1).

1.1 Change in HALE between 2000 and 2019

1.1.1 Global

Globally, the 5.4-year increase in HALE at birth between 2000 (58.1 years) and 2019 (63.5 years) was a result of declining mortality and morbidity over time, contributing 5.2 years (96.0%) and 0.2 years (4%), respectively. Reduction in mortality from communicable, maternal, perinatal and nutritional conditions (referred to as communicable diseases) represents the greatest source of gain, contributing a total of 3.4 years to the HALE gain, while mortality reduction from NCDs contributed 1.4 years, and injuries 0.4 years (Fig. 1.2).

By individual cause and broad age groups, the largest gains were made by mortality reduction in preterm birth complications (0.26 years) and birth asphyxia and birth trauma (0.21 years) among infants aged 0–1 years, lower respiratory infections (0.40 years), diarrhoeal diseases (0.35 years) and measles (0.24 years) among children aged 0–4 years, and tuberculosis (0.27 years) and HIV/AIDS (0.22 years) among adults aged 30–69 years. The gain attributed to NCD mortality reduction was concentrated in the adult age groups, with stroke contributing 0.37 years, ischaemic heart disease 0.28 years and chronic obstructive pulmonary disease 0.19 years among adults 30 years and older. While the majority of causes contributed to the HALE gain positively, increasing mortality linked to Alzheimer disease and other dementias among adults aged 70 years and older resulted in a slight two-week (0.04-year) HALE loss in 2000–2019 (Fig. 1.2).

While the overall contribution to the HALE gain from changing morbidity in 2000–2019 was relatively minor and, for the majority of causes, the contribution was positive, it is worth noting that morbidity from diabetes among adults 30 years and older increased and was responsible for a 0.14-year HALE loss in 2000–2019 (Fig. 1.2).

Figure 1.2 Decomposition of the change in healthy life expectancy (HALE) between 2000 and 2019 globally, by cause and age group, for both sexes combined



1.1.2 WHO regions

As the cause and the age profile of mortality and morbidity vary across geographical locations, the contributions to HALE by cause and age also exhibit diverse patterns.

In the WHO African Region, the most remarkable contributor to the HALE gain in 2000–2019 was from reduction in HIV/AIDS mortality, resulting in a total

of a 2.61-year gain in HALE, within which a reduction among adults aged 30–69 years was the major source (1.71 years). Decline in mortality from tuberculosis in the same age group also contributed to a large share (0.68 years or 57.6%) of the 1.18-year HALE gain accumulated throughout the entire life course. Mortality decline among children aged under 5 years from malaria, diarrhoeal diseases and lower respiratory infections also contributed considerably to the HALE gain in 2000–2019, at 0.88, 0.65 and 0.62 years, respectively (Fig. 1.3).

Figure 1.3 Leading contributing causes for change in healthy life expectancy (HALE) between 2000 and 2019, by cause, age group and WHO region, for both sexes combined



AFR: African Region; AMR; Region of the Americas; SEAR: South-East Asia Region; EUR: European Region; EMR: Eastern Mediterranean Region; WPR: Western Pacific Region.

Source: WHO (1).

Similarly, in the Eastern Mediterranean Region and the South-East Asia Region, reduction in mortality from communicable diseases was the primary driver of the HALE gain in 2000–2019. Diarrhoeal diseases contributed a total of 1.01 years (0.47 years among those aged under 5 years) in HALE gain in the South-East Asia Region and 0.51 years (0.38 years among those aged under 5 years) in the Eastern Mediterranean Region. A mortality decline in lower respiratory infections among children aged under 5 years led to HALE gains of 0.57 years in the South-East Asia Region and 0.38 years in the Eastern Mediterranean Region. While the Eastern Mediterranean Region enjoyed a 0.26-year gain in HALE due to declining mortality from ischaemic heart disease among adults aged 30 years and older, the South-East Asia Region saw an opposite trend, with rising mortality from the diseases causing 0.16-year HALE loss among the same age groups. Also worth noting is the 0.3-year loss attributed to increasing diabetes morbidity among those aged 30 years and older in the Eastern Mediterranean Region, constituting the largest loss among all WHO regions. The Eastern Mediterranean Region is also the only region where a sizable HALE loss was attributed to mortality increase in collective violence and legal intervention, at 0.28 years, of which 0.24 years was due to rising mortality among those aged 5-29 years (Fig. 1.3).

In contrast, in the Region of the Americas the contribution to the gain in HALE was more heavily concentrated in NCDs, with declining mortality among adults aged 30 years and older from ischaemic heart disease (0.61 years), stroke (0.24 years), and trachea, bronchus and lung cancers (0.16 years) driving the gain (Fig. 1.3).

However, some negative patterns were also observed. Rising mortality due to Alzheimer disease and other dementias among adults aged 70 years and older was responsible for a 0.12-year HALE loss. While improvement in mortality from diabetes contributed to a minor increase (0.03 years) in HALE in 2000–2019, worsening morbidity from the disease caused a 0.22-year loss. Additionally, drug use disorder in the Region of the Americas contributed negatively through increasing both mortality and morbidity, causing 0.11-year and 0.13-year losses, respectively, a unique phenomenon compared with all other WHO regions (Fig. 1.3).

In the European Region, there was a 4.69-year gain in HALE between 2000 and 2019, with 93.5% (or 4.39 years) of this gain attributed to declining mortality. By far, the leading driver in HALE gain was the decline in mortality from ischaemic heart disease among adults aged 30 years and older, accounting for a 1.07-year HALE gain, or over one fifth of the total gain. This is followed by mortality reduction from stroke among the same age groups, contributing to a further 0.66-year increase in HALE. The European Region also benefitted from reduced mortality from lower respiratory infections, road injuries and suicide across many age groups over the life course, achieving a HALE gain of about 0.2 years for each cause (0.6 years in total). However, increasing mortality from Alzheimer disease and other dementias among adults aged 70 years and older and increasing morbidity from diabetes led to a loss in HALE, at 0.08 and 0.14 years respectively (Fig. 1.3).

In the Western Pacific region, HALE increased by 4.53 years between 2000 and 2019, about 97% (or 4.43 years) of which was achieved by reducing mortality. The most prominent HALE gain was achieved through reducing mortality from stroke and chronic obstructive pulmonary disease among adults aged 30 years and older, accounting for an increase of 0.64 and 0.62 years, respectively. Declining mortality from birth asphyxia and birth trauma, preterm birth complications and lower respiratory infections among children aged under 1 year led to an increase in HALE of around 0.25 years for each of the three diseases (0.76 years in total). As in other regions, diabetes-related morbidity increased among adults aged 30 years and older and caused a 0.1-year loss in HALE (Fig. 1.3).

1.2 Change in HALE during the COVID-19 pandemic in 2019–2021

1.2.1 Change attributed to mortality

The Global HALE dropped by 1.54 years between 2019 and 2021. Mortality from the emerging COVID-19 and other outcomes related to the COVID-19 pandemic led to 1.29-year and 0.28-year losses, respectively, in HALE, totalling a staggering 1.57-year loss within just two years and exceeding the total gains in reduced mortality and morbidity from other causes. The majority of the loss was concentrated among those aged 30 years and older, at 1.23 years and 0.27 years, respectively, attributed to the two causes (Fig. 1.4).

While gain due to continued decline in mortality from causes including HIV/AIDS and tuberculosis was observed in 2019–2021 in the African Region, the region suffered from a net loss of 0.66 years in HALE. The loss is predominantly driven by mortality from COVID-19 (0.74 years) and other COVID–19 pandemic-related outcomes (0.29 years) among adults aged 30 years and older (Fig. 1.4).





AFR: African Region; AMR; Region of the Americas; SEAR: South-East Asia Region; EUR: European Region; EMR: Eastern Mediterranean Region; WPR: Western Pacific Region.

Source: WHO (1).

Much of the 2.72-year HALE loss in the Region of the Americas can be attributed to COVID-19 mortality among adults aged 30 years and older (1.92 years) and morbidity for all ages (0.15 years). Loss in HALE can also be explained by causes for which the previous pre-pandemic progress in mortality reduction was completely reversed or halted, including ischaemic heart disease, which contributed negatively to the HALE change between 2019 and 2021 (-0.07 years in 2019–2021 vs 0.61 years in 2000–2019) and stroke, which contributed almost no gain between 2019 and 2021 (compared with a 0.24-year gain in 2000–2019) (Fig. 1.4).

Mortality from COVID-19 and other COVID-19 pandemicrelated outcomes among adults aged 30 years and older contributed 2.05 and 0.56 years, respectively, to the total 2.5-year HALE loss in the South-East Asia Region between 2019 and 2021. This outweighed HALE gains achieved through mortality reductions from causes including ischaemic heart disease (0.09 years) and stroke (0.08 years) among those aged 30 years and older during this two-year period (Fig. 1.4).

The overall HALE dropped by 1.5 years in 2019–2021 in both the European Region (66.1 to 64.6 years) and the Eastern Mediterranean Region (60.4 to 58.9 years), and mortality from COVID-19 and other COVID-19 pandemicrelated outcomes among adults aged 30 years and older accounted for 1.35 and 0.25 years, respectively, of the loss in the European Region and 1.31 and 0.46 years, respectively, in the Eastern Mediterranean Region (Fig. 1.4).

The COVID-19 pandemic had a very limited impact on HALE in the Western Pacific region in 2020 and 2021, resulting in virtually no change in HALE (68.4 years in 2019 and 68.2 years in 2021). Mortality from COVID-19 and other COVID-19 pandemic-related outcomes only accounted for 0.09 and 0.02 years of loss, respectively, in HALE, which were compensated for by minor gains from mortality decline from other causes, including stroke and road injuries (Fig. 1.4).

1.2.2 Change attributed to morbidity

The decline in HALE between 2019 and 2021 at global and regional levels can be partly attributed to increasing morbidity. A shared pattern at both global and regional levels is that the HALE loss in 2019–2021 associated with morbidity increase can largely be attributed to three causes: COVID-19, anxiety disorders and depressive disorders. It is also worth noting that the pre-pandemic HALE gains in 2000–2019 attributable to declining morbidity from anxiety disorders and depressive disorders were completely cancelled out by the subsequent loss associated with these two causes during the first two years of the COVID-19 pandemic (Fig. 1.4).

Globally, morbidity due to COVID-19 led to a 0.13-year HALE loss in 2019-2021. The pandemic also caused increasing morbidity due to anxiety disorders and depressive disorders, each responsible for a 0.06-year (or 3-week) loss in HALE in 2019–2021. This loss (0.12 years, or 6 weeks) effectively wiped out 80% of the positive contribution to HALE (0.15 years, or approximately 8 weeks) by mortality decline from all NCDs combined in that two-year period. By WHO regions, morbidity due to COVID-19 led to HALE loss in 2019–2021 ranging from 0.01 years in the Western Pacific region to 0.18–0.20 years in the South-East Asia Region, the Eastern Mediterranean Region and the African Region. Increasing morbidity due to anxiety disorders was responsible for losses ranging from 0.03 years in the African Region to 0.10 years in the Region of the Americas. Similarly, the contribution to overall HALE loss in 2019–2021 from increasing morbidity due to depressive disorders ranged from 0.01 years in the Western Pacific region to 0.11 years in the Region of the Americas (Fig. 1.4).

1.3 Gap in HALE between males and females in 2019 and 2021

1.3.1 Global

On average, women live longer than men, as they benefit from lower mortality, especially from causes that are affected by behavioural or genetic factors (8,9,10,11). In 2019, just before the COVID-19 pandemic, the global female life expectancy was about 5.1 years longer than that of males (75.70 versus 70.61 years). However, with higher overall risk of morbidity among females, when disability is accounted for, the gap between the global male and female HALE in 2019 reduced to about 2.3 years (8,10). The lower level of mortality among females would have rendered them a 3.85-year advantage in HALE, yet they also faced higher levels of morbidity than males, offsetting the HALE advantage associated with mortality by 1.56 years (Fig. 1.5). Lower female mortality from NCDs among adults aged 30 years and older from ischaemic heart disease (accounting for 0.63 years), stroke (0.31 years), chronic obstructive pulmonary disease (0.26 years), and trachea, bronchus and lung cancers (0.24 years) are major contributors to the female advantage in HALE. In addition, lower mortality among those aged 5–69 years old from injuries including road injury (0.32 years), interpersonal violence (0.17 years) and suicide (0.12 years) constitute an important source of the female HALE advantage. However, mortality from sex-specific causes, such as breast cancer and maternal conditions, reduced the HALE advantage for females, at 0.22 and 0.15 years, respectively (Fig. 1.5).



Figure 1.5 Leading contributing causes for the difference in healthy life expectancy (HALE) in 2019 between males and females, by cause and age group

AFR: African Region; AMR; Region of the Americas; SEAR: South-East Asia Region; EUR: European Region; EMR: Eastern Mediterranean Region; WPR: Western Pacific Region. Source: WHO (1). In addition, a higher rate of morbidity from many conditions has cost more healthy life-years among females. Back and neck pain, gynaecological diseases and migraine among females aged 30 years and older reduced female HALE by 0.39 years, 0.38 years and 0.13 years, respectively, compared with males, and thus effectively eliminated the HALE advantage associated with mortality due to ischaemic heart disease and chronic obstructive pulmonary disease. Furthermore, depressive disorders and anxiety disorders reduced the overall additional HALE for females by 0.20 and 0.16 years, respectively (Fig. 1.5).

In 2021, HALE for females was about 2.2 years higher than for males, representing the net of a 3.8-year advantage

associated with mortality and a 1.7-year disadvantage associated with morbidity. In general, the mortality and morbidity contributors to the female–male gap in HALE remained largely the same, although the size of each contribution diminished to different degrees. What is most noteworthy is that lower female mortality from COVID-19 (0.72 years) and other COVID-19 pandemicrelated outcomes (0.17 years) have contributed to a 0.89-year HALE advantage over males globally. This contribution is dominated by those aged 30 years and older (0.70 and 0.16 years, respectively, for COVID-19 and other COVID-19 pandemic-related outcomes), a pattern that also holds true at regional and country levels (Fig. 1.6).

Figure 1.6 Leading contributing causes for the difference in healthy life expectancy (HALE) in 2021 between males and females, by cause and age group



AFR: African Region; AMR; Region of the Americas; SEAR: South-East Asia Region; EUR: European Region; EMR: Eastern Mediterranean Region; WPR: Western Pacific Region.

1.3.2 WHO regions

Similar patterns were observed across WHO regions, yet given each region's unique epidemiological profile, there are distinctive characteristics in the health inequality between male and females in each region. In the African Region, sex-difference in mortality from communicable diseases was the primary source of the HALE gap (1.35 years) between sexes in 2019. Lower tuberculosis mortality among adults aged 30 years and older brought a 0.48-year HALE advantage to females. Lower mortality from lower respiratory infections among children aged under 5 years and adults aged 30 years and older contributed a further 0.30 years. However, these gains were partially offset by mortality from maternal conditions in reproductive ages, which cost females 0.54 years in HALE (Fig. 1.5). In 2021, lower female mortality from COVID-19 and other COVID-19 pandemicrelated outcomes accounted for an additional 0.47 and 0.18 years, respectively, in female HALE over male. This is predominantly contributed by females aged 30 years and older, accounting for 0.46 and 0.18 years, respectively, for the two causes (Fig. 1.6).

A unique contributor to the female HALE lead (2.49 years) in 2019 in the Region of the Americas was mortality due to interpersonal violence, accounting for 0.71 years of the HALE gap and predominantly concentrated in those aged 5–69 years (Fig. 1.5). In 2021, lower female mortality from COVID-19 and other COVID-19 pandemic-related outcomes explained 0.96 and 0.03 years, respectively, of the lead in female HALE (2.96 years in total) (Fig. 1.6).

Stroke is the most significant cause of death that contributed to the female HALE lead (3.43 years in total) in the Western Pacific region in 2019, accounting for 0.73 years (Fig. 1.5). In 2021, mortality from COVID-19 and other COVID-19 pandemic-related outcomes had very limited impact on the HALE gap between males and females, accounting for only 0.03 and 0.01 years, respectively (Fig. 1.6).

The Eastern Mediterranean Region had the narrowest male–female gap in HALE in 2019, at 0.35 years. Mortality from collective violence and legal intervention stands out as a prominent driver for the HALE gap compared with other regions, contributing 0.45 years to the female HALE lead, with fatality among those aged 5–29 years accounting for over 90% of it (0.41 years). The relatively narrow overall HALE gap can be partially explained by large contributions from some conditions for which females faced higher mortality or morbidity, particularly morbidity from gynaecological diseases (-0.71 years) with women aged 20–69 years accounting for -0.66 years of this (Fig. 1.5). In 2021, the female lead in HALE declined slightly to 0.33 years, despite the addition of major contributions associated with mortality from COVID-19 (0.61 years) and other COVID-19 pandemicrelated outcomes (0.3 years). This was due to declining contributions from mortality from causes including collective violence and legal intervention (down from 0.45 years in 2019 to 0.28 years in 2021) and ischaemic heart disease (down from 0.49 years in 2019 to 0.34 years in 2021) (Fig. 1.6).

The South-East Asia Region had the second lowest male-female gap in HALE in 2019 after the Eastern Mediterranean Region, at 0.90 years. Females benefitted from lower mortality from ischaemic heart disease, which accounted for 0.76 years of the female HALE advantage, primarily concentrated in those aged 30 years and older (0.75 years) (Fig. 1.5). The overall HALE gap remained relatively constant in 2021, at 0.91 years. Lower female mortality from COVID-19 and other COVID-19 pandemicrelated outcomes accounted for 1.28 and 0.33 years, respectively, of the lead in female HALE. It is worth noting that the contribution associated with ischaemic heart disease halved, to 0.37 years, compared with 2019, indicating that as a competing risk the emerging deaths attributed to the COVID-19 pandemic have narrowed the male-female mortality gap associated with ischaemic heart disease (Fig. 1.6).

Females in the European Region enjoyed a 2.98-year HALE lead in 2019 compared with males. Lower female mortality from ischaemic heart disease among adults aged 30 years and older represents the greatest contributor, accounting for a total of 0.91 years. Lower female mortality from trachea, bronchus and lung cancers in the same age group adds another 0.44 years to the female HALE lead (Fig. 1.5). In 2021, the malefemale HALE gap dropped to 2.76 years, with lower female mortality from COVID-19 and other COVID-19 pandemic-related outcomes accounting for 0.65 and 0.11 years, respectively, and the contribution associated with mortality from ischaemic heart disease (down to 0.75 years) and trachea, bronchus and lung cancers declining from the 2019 levels (down to 0.36 years). Contributions from all four causes to the female HALE lead were concentrated in those aged 30 years and older (Fig. 1.6).

1.4 Gap in HALE between highincome and other income groups in 2019 and 2021

Populations living in lower-resourced settings face many socioeconomic challenges and higher health risks than those living in higher-resourced settings (12), resulting in disproportionately higher morbidity and mortality from many causes and poorer overall health outcome as measured by life expectancy and HALE (1).

When looking at HALE in high-income countries compared with other income groups (as classified by the World Bank), the main health issues that explain the differences – and how much they contribute – vary depending on which groups are being compared.

In 2019, people in high-income countries lived longer healthy lives partly because they had lower death rates from certain diseases and conditions, including birth asphyxia and birth trauma, preterm birth complications, lower respiratory infections, diarrhoeal diseases and tuberculosis. These factors explained 4.08 years of the 12.62-year HALE gap between high- and low-income countries, and 3.08 years of the 8.61-year gap between high- and lower-middle-income countries.

However, these same health issues only explained a small part – about 0.38 years – of the 2.08-year HALE difference between high- and upper-middle-income countries.

Other major health conditions that contributed to the HALE gap were ischaemic heart disease and stroke. Lower death rates from these conditions in high-income countries accounted for: 0.52, 0.95 and 0.4 years of the HALE gap due to ischaemic heart disease, and 1.14, 0.76 and 1.01 years due to stroke, in comparisons with low-, lower-middle- and upper-middle-income countries, respectively (Fig. 1.7).

Morbidity Mortality 0.02 0.01 0 09 0.26 1. Lower respiratory infections -0.02 0.01 0.05 0.03 3. Tuberculosi: 0.03 0.05 -0.01 0.15 0.07 2. Depressive disorders 0.24 0.16 0.07 0.03 0.2 0.16 4. Diarrhoeal dise -ow-income 0.25).32 Malaria 0.12 0.05 0.02 0.02 -0.03 -0.04 -0.03 3. Back and neck pain 6. Preterm birth complications 0.28 0.23 8 Ischaemic heart disea -0.01 -0.08 -0.11 4. Drug use disorders Birth asphyxia and birth trauma -0.07 -0.1 0.01 0.03 0.12 -0.03 -0.08 0.02 1. Ischaemic heart disease -0.03 -0.05 -0.07 1. Back and neck pain 2. Stroke ncom piratory infections 0.04 0.11 0.11 -0.01 -0.08 0.19 0.11 0.04 0.01 0.13 0.24 4. Diarrhoeal diseases iddle-0.23 -0.01 -0.1 -0.04 Tuberculosi 0.04 0.04 0.03 0.14 0.04 7. Preterm birth complications 0.06 0.09 4. Chronic obstructive pulmonary disease Birth asphyxia and birth trauma OWe -0.09 -0.03 -0.03 -0.06 -0.01 -0.03 -0.09 17. Drug use disorders 1. Back and neck pair -0.03 -0.04 -0.07 Ischaemic heart disease Chronic obstructive pulmonary disease middle-income 0.05 -0.1 -0.1 0.03 0.03 Road injury Stomach cancer 0.06 .08 -0.01 -0.07 -0.1 -0.03 -0.1 0.01 0.03 8. Tuberculosis 0.03 0.04 0.11 0.07 10. Preterm birth complications Joper 15. Birth asphyxia and birth trauma Trachea, bronchus, lung can Diarrhoeal disease -0.02 -0.05 0.03 0.01 0.01 0 1-4 5-19 20-29 30-69 1-4 5-19 20-29 30-69 70+70 +Age Difference in HALE (years) Cause group All causes Communicable, maternal, perinatal and nutritional conditions 0.50 -0.25 0.00 0.25 0.50 Injuries Noncommunicable diseases

Figure 1.7 Leading contributing causes for the difference in healthy life expectancy (HALE) in 2019 between highincome and other income groups, by cause and age group

However, high-income countries did not succeed in every aspect. Higher mortality due to trachea, bronchus and lung cancers, particularly among adults aged 30 years and older, led to a negative contribution to the difference in HALE when comparing high-income countries against low-income (-0.17 years) and lower-middle-income countries (-0.2 years). Additionally, populations in highincome countries on average faced higher mortality from drug use disorders in comparison with all the other three income groups, costing high-income countries 0.12, 0.13 and 0.13 years of the HALE lead, respectively, compared with low-income, lower-middle-income and uppermiddle-income countries, respectively (Fig. 1.7).

High-income countries also sustained higher rates of morbidity due to some injuries and NCDs. While overall morbidity in high-income countries was still lower than that in low- and lower-middle-income countries in 2019, it was higher than upper-middle-income countries and led to a nearly 1-year negative contribution to the HALE gap between high- and upper-middle-income countries. The major causes leading to high-income countries' disadvantage include falls (losing 0.32, 0.16 and 0.21 years compared with low-, lower-middle- and uppermiddle-income countries, respectively), back and neck pain (-0.21, -0.38 and -0.36 years, respectively) and drug use disorders (-0.20, -0.20 and -0.19 years, respectively) (Fig. 1.7).

These patterns in general continued in 2021. However, the different impacts of the COVID-19 pandemic in individual income groups has influenced the gaps in HALE to various degrees compared with high-income countries. Lower mortality due to COVID-19 and other COVID-19 pandemic-related outcomes in high-income countries accounted for the HALE lead of 0.25 years and 0.32 years, respectively, compared with low-income countries; 1.39 and 0.47 years, respectively, compared with lower-middle-income countries; and 0.21 and 0.14 years, respectively, compared with upper-middle-income countries (Fig. 1.8).









Source: WHO (1).

Injuries

Noncommunicable diseases

Other COVID-19 pandemic-related outcomes

1.5 Progress in premature mortality reduction

Deaths occurring before the age of 70 years are considered premature and the majority of these deaths are preventable. Taking into account of the changing population structure over time and its difference across regions, the age-standardized death rate (ASR) for ages under 70 years was in steady decline from 2000 until the start of the COVID-19 pandemic in 2019, both globally and at regional level. It is evident that before the COVID-19 pandemic, NCDs were making up an increasing share of the ASRs of all deaths under age of 70 years across all WHO regions, while the share of communicable diseases was in decline (Fig. 1.9). In 2019, the African Region had the highest ASR of all deaths under age of 70 years (665 per 100 000 population) among all WHO regions, over 80% higher than the global average (366 per 100 000 population) and nearly triple that of the Western Pacific Region (233 per 100 000), the region with the lowest ASRs. The African Region also had the largest share of communicable diseases in the overall ASR under age of 70 years in 2019, at 47.4%, some 5.5 times of that in the European Region (8.6%), which had the lowest share (Fig. 1.9).

Figure 1.9 Composition of causes of death in the age-standardized death rates for ages under 70 years, by WHO region, 2000–2021



Source: WHO (1).

The probability of dying before the age of 70 years is another important indicator for overall population health with a focus on mortality. Premature mortality had been in decline globally and in all WHO regions, from 2000 to the onset of the COVID-19 pandemic. Globally, a newborn had about 40.4% chance of dying before reaching the age of 70 years in 2000, declining to 29.9% in 2019. Health disparities associated with geography and income are also significant. Africa, the WHO region with the highest premature mortality in 2019, had a probability (46.3%) of dying before the age of 70 years that was double that of the European Region (23.2%) and the Western Pacific region (21.5%), which had the lowest probability. Similarly, the probability in low-income countries (46.8%) was also more than double that of high-income countries (19.8%).

The COVID-19 pandemic has had a marked impact on global health, with global premature mortality increasing by 15.0% between 2019 (29.9%) and 2021 (34.3%). The pandemic impacts were unequal across WHO regions and World Bank income groups. The mortality toll due to COVID-19 was heaviest in relative terms in the Region of the Americas and the South-East Asia Region, where premature mortality increased by 28.9% and 23.7%, respectively. In contrast, the pandemic had a limited impact in the African Region and the Western Pacific Region, where premature mortality increased by just 5.7% and 2.0%, respectively, between 2019 and 2021. By World Bank income groups, the premature mortality in lowincome countries increased modestly, by just 5%, during the first two years of the pandemic, while all other income groups experienced a 14.3-17.8% increase.

Despite the overall progress pre-pandemic, stagnation was seen after 2015. The global annual rate of reduction (ARR) in premature mortality was just 1.1% between 2015 and 2019, down from 1.7% during 2000-2015. Among WHO regions, the Region of the Americas and the South-East Asia Region experienced the greatest slowdown, with ARR halving between 2015 and 2019 compared with 2000–2015. The Eastern Mediterranean Region was the only region where some acceleration was seen in 2015-2019, while the progress in the African Region slowed only slightly and the region continued to have the most rapid rate of decline in premature mortality, followed by the European Region. Similarly, progress slowed in 2015-2019 across all World Bank income groups, with lowincome countries seeing the smallest relative reduction in ARR and remaining the group with the fastest declining premature mortality.

1.5.1 The prospect of premature mortality by 2050

Using the WHO mortality estimates in 2019 as the baseline and incorporating the projected age-sexspecific mortality trends by country from the United Nations World population prospects 2024 (13), it is anticipated that the reduction in premature mortality during the three decades between 2019 and 2050 would be moderate compared with the observed progress in 2000-2019. Globally, the probability of dying before the age of 70 years is projected to decline by 24.2% between 2019 and 2050, with reductions ranging from 20.2% in the African Region and 24.5% in the Eastern Mediterranean Region to 33.9% in the Western Pacific region and 35.4% in the South-East Asia Region. This represents a global slowdown, given it had taken one fewer decade to achieve a slightly larger reduction (26.1%) in 2000–2019, equivalent to an estimated ARR of 1.6%, compared with only a projected 0.9% in 2019–2050. Overall, only about one third of countries are projected to have accelerated progress in 2019–2050 compared with 2000–2019. This would largely occur in the Region of the Americas and the Western Pacific Region, where at least half of the countries (76% and 52%, respectively) would see an acceleration (Fig. 1.10). Upper-middle-income countries are the only World Bank income group where more than half (52%) of the countries are projected to see accelerated progress in 2019-2050, while only four (or 15%) of the low-income countries will achieve this (Fig. 1.10).

The Region of the Americas is the only WHO region that is expected to see some accelerated progress, with a projected ARR at 1.2% in 2019-2050, up from 0.9% in 2000-2019. While the ARR in the South-East Asia Region is expected to be largely stable, the other four regions are all expected to see considerable deceleration, with ARR dropping by up to 60% in the African Region. Slowed progress is projected for all World Bank income groups. Although a one-third reduction in premature mortality is expected in both upper-middle-income and highincome countries in 2019–2050, the ARRs are considerably lower than that in 2000-2019 (down by 31.4% and 19.5%, respectively). The ARR in low-income countries is expected to be more than halved the level in 2000-2019, leading to only about a one-fifth reduction in premature mortality in 2019-2050.

Should the global community wish to halve premature mortality by 2050 (14), considerable acceleration would be needed in reducing mortality rates in those aged 0–69 years in the next three decades. The global decline would have to be 1.4 times the observed rate in 2000–2019 to cut the premature mortality in 2019 by half by mid-century. The required regional rate of acceleration relative to the ARR in 2000–2019 would range from 1.1 times in the European Region and the Western Pacific region to 2.4 times in the Region of the Americas, and from 1.2 times in upper-middle-income countries to 1.6 times in lower-middle-income countries. Allowing the mortality trends to continue as projected without interventions would shift a heavier burden of acceleration for meeting the target to future years. At the projected ARR for 2019–2050, a minimum of 2.5 times acceleration is required to halve premature mortality by 2050 globally, and a minimum of about a threefold acceleration in the African Region and low-income countries.

Figure 1.10 Annual rate of reduction in probability of death under 70 years of age, observed (2000–2019) versus projected (2019–2050)





Source: WHO (1).

References¹

- 1. Global health estimates 2021 [website]. Geneva: World Health Organization; 2024 (https://www.who.int/data/global-healthestimates, accessed on 08 January 2025).
- Delivery care: UNICEF/WHO joint database on births attended by skilled health personnel. New York: United Nations Children's Fund and World Health Organization; 2024 (https:// data.unicef.org/topic/maternal-health/delivery-care/, accessed on 15 February 2025).
- 3. Global report on hypertension: the race against a silent killer. Geneva: World Health Organization; 2023 (https://iris.who.int/ handle/10665/372896). Licence: CC BY-NC-SA 3.0 IGO.
- 4. 2024 global AIDS report the urgency of now: AIDS at a crossroads. UNAIDS global AIDS update 2024. Geneva: Joint United Nations Programme on HIV/AIDS; 2024.
- United Nations Inter-agency Group for Child Mortality Estimation (UN IGME). Levels & trends in child mortality: report 2024 – estimates developed by the United Nations Inter-agency Group for Child Mortality Estimation. New York: United Nations Children's Fund; 2025.
- Andreev EM, Shkolnikov VM, Begun AZ. Algorithm for decomposition of differences between aggregate demographic measures and its application to life expectancies, healthy life expectancies, parity-progression ratios and total fertility rates. Demogr Res. 2002;7:499–522 (https://doi.org/10.4054/demres.2002.7.14).
- Beltrán-Sánchez H, Preston SH, Canudas-Romo V. An integrated approach to cause-of-death analysis: causedeleted life tables and decompositions of life expectancy. Demogr Res. 2008;19:1323–1350 (https://doi.org/10.4054/ DemRes.2008.19.35).

- Case A, Paxson C. Sex differences in morbidity and mortality. Demography. 2005;42(2):189–214 (https://doi.org/10.1353/ dem.2005.0011).
- Waldron I. Sex differences in human mortality: the role of genetic factors. Soc Sci Med. 1983;17(6):321–333 (https://doi. org/10.1016/0277-9536(83)90234-4).
- Crimmins EM, Shim H, Zhang YS, Kim JK. Differences between men and women in mortality and the health dimensions of the morbidity process. Clin Chem. 2019;65(1):135–145 (https://doi.org/10.1373/clinchem.2018.288332).
- Rogers RG, Everett BG, Saint Onge JM, Krueger PM. Social, behavioral, and biological factors, and sex differences in mortality. Demography. 2010;47(3):555–578 (https://doi. org/10.1353/dem.0.0119).
- Marmot M. Social determinants of health inequalities. Lancet. 2005;365(9464):1099–1104 (https://doi.org/10.1016/S0140-6736(05)71146-6).
- United Nations Department of Economic and Social Affairs Population Division. World population prospects 2024. New York: United Nations; 2024 (https://population.un.org/wpp/).
- Jamison DT, Summers LH, Chang AY, Karlsson O., Mao W, Norheim OF et al. Global health 2050: the path to halving premature death by mid-century. Lancet. 2024;404(10462):1561–1614 (https://doi.org/10.1016/S0140-6736(24)01439-9).

Unless otherwise stated, all references accessed on 5 May 2025.

Health-related Sustainable Development Goals

This chapter presents the latest available evidence of trends in health-related SDG indicators, assessing progress towards achieving the global targets. The topics covered in this chapter include child mortality, cause-specific mortality, infectious diseases, risk factors for health, and UHC and health systems.

2.1 Mortality-related Sustainable Development Goal indicators

2.1.1 Maternal and child mortality

Maternal mortality

Globally, an estimated 260 000 (uncertainty interval [UI]: 230 000-309 000) women died in 2023 due to maternal conditions, equivalent to an overall maternal mortality ratio (MMR; SDG indicator 3.1.1) of 197 (UI: 174-234) maternal deaths per 100 000 live births. This represents a decline in the global MMR by 40% since 2000, when the rate was 328 (UI: 308-352) per 100 000 live births. Between 2000 and 2023, the average ARR in global MMR was 2.2% (UI: 1.4-2.8%). This led to a significant fall in the number of maternal deaths globally, from 444 000 in 2000 to 260 000 in 2023. A closer examination of the first eight years of the SDG era, between the beginning of 2016 and the end of 2023, reveals that the global MMR declined by 10%, from an estimated 220 (UI: 203-242) maternal deaths per 100 000 live births, at a global average ARR of 1.6% (UI: 0.04-2.7%) (1). If this ARR were to continue until

2030, the global MMR would be 177 in 2030 – two and a half times the SDG target (below 70) *(1,2)*. Achieving the SDG target will require an average ARR of 14.8% over the remaining seven years (2024–2030). This represents an unprecedented challenge and would be equivalent to almost 700 000 deaths averted between 2024 and 2030, compared with a scenario where the 2016–2023 global ARR continues.

MMR declined in all WHO regions between 2000 and 2023. The greatest decline was achieved in the South-East Asia Region, where MMR declined by 73.6%, from 365 (UI: 336–407) in 2000 to 96 (UI: 86–112) maternal deaths per 100 000 live births in 2023 (Fig. 2.1). This equates to an average ARR of 5.8% (UI: 5.1–6.5%). MMR also declined by 56.0% in the Eastern Mediterranean Region, 55.2% in the European Region, 51.2% in the Western Pacific Region, and 39.6% in the African Region between 2000 and 2023. The smallest percentage reduction in MMR was in the Region of the Americas at 16.9%, equivalent to an ARR of 0.8% (UI: 0.2–1.3%).

Figure 2.1 Estimates of maternal mortality ratio (per 100 000 live births) 2000–2023, by WHO region



Source: WHO et al. (1).

In 2023, the African Region had a high MMR of 442 (UI: 376–560) per 100 000 live births and alone accounted for approximately 70% of global maternal deaths in 2023. The Eastern Mediterranean and South-East Asia Regions each accounted for about 12% of global maternal deaths, with MMRs of 167 (UI: 135–218) and 96 (UI: 86–112), respectively. The MMR was also low in the Region of the Americas, at 59 (UI: 52–66), and the Western Pacific Region, at 35 (UI: 30–42). The European Region had a very low MMR in 2023, at 11 (UI: 10–13) (Fig. 2.1) (1).

Under-five mortality

Child survival has improved remarkably since 2000, with the global under-five mortality rate (U5MR; SDG indicator 3.2.1) declining by more than half (52%) in the last two decades, from 77 (UI: 76–78) deaths per 1000 live births in 2000 to 37 (UI: 35–41) deaths per 1000 live births in 2023 (Fig. 2.2). Accordingly, the number of global under-five deaths fell to 4.8 million (UI: 4.5–5.3 million) in 2023, a number that is still unacceptably high, despite accounting for less than half of the under-five deaths in 2000 (10.1 million [UI: 9.9–10.2 million]) and almost two thirds less than the estimated 13 million (UI: 12.8– 13.2 million) in 1990 (*3*). Although the progress is noteworthy at global level, stark inequalities persist across regions and income levels. The WHO African Region remains the region with the highest U5MR, estimated at 67 (UI: 60–80) deaths per 1000 live births in 2023 (Fig. 2.2). The African Region alone accounted for 55% (2.6 million [UI: 2.4–3.1 million]) of the global deaths of children under age 5 in 2023. In 2023, the U5MR in low-income countries (LICs) (62 [UI: 56–75] deaths per 1000 live births) and lower-middle-income countries (LMICs) (43 [UI: 39–50] deaths per 1000 live births) was, respectively, 13 and 9 times the rate in highincome countries (HICs) (5 [UI: 5–5] deaths per 1000 live births) (*3*).

Globally, infectious diseases, including acute respiratory infections, malaria and diarrhoea, along with pre-term birth complications, birth asphyxia and trauma, and congenital anomalies remain the leading causes of death for children aged under 5. Access to basic lifesaving interventions such as skilled delivery at birth, quality postnatal care, early and continued breastfeeding and adequate nutrition, immunization and treatment for common childhood diseases can save many young lives (3).

Figure 2.2 Child mortality rate (per 1000 live births), by WHO region, 1990, 2000 and 2023



Source: UN IGME (3).

Neonatal mortality

The number of deaths occurring in the first 28 days of life (the newborn period) declined from 4.2 million (UI: 4.0–4.3 million) in 2000 to 2.3 million (UI: 2.1–2.6 million) in 2023. However, the 45% decline in the neonatal mortality rate (NMR; SDG indicator 3.2.2) from 2000 (31 [UI: 30–32] deaths per 1000 live births) to 2023 (17 [UI: 16–19] deaths per 1000 live births), has been slower compared with the 58% decline in the mortality rate of children aged 1–59 months (from 47 [UI: 47–48] deaths per 1000 children aged 28 days in 2000 to 20 [UI: 18–23] deaths per 1000 children aged 28 days in 2023) (Fig. 2.2). This has led to a larger share of global newborn deaths among total under-five deaths, up from 41% in 2000 to 48% in 2023 (*3*).

The chances of survival through the first 28 days of life vary depending on where a child is born. The African Region had the highest regional NMR in 2023, at 26 (UI: 23–31) deaths per 1000 live births, followed by the Eastern Mediterranean Region with 24 (UI: 21–29) deaths per 1000 live births. The South-East Asia Region had the highest regional NMR in 1990 at 53 (UI: 51–55) deaths per 1000 live births (Fig. 2.2), but has experienced a sharp decline with a ARR of 3.6% (UI: 3.2–4.0%) double that of the African Region (1.7% [UI: 1.1–2.0%]) in 1990–2023. Country-level NMR in 2023 ranged from 0.6 to 40.2 deaths per 1000 live births and the risk of dying before the 28th day of life for a child born in the highest-mortality country was approximately 65 times higher than the lowest-mortality country (*3*).

Meeting the SDG targets for child survival

If current trends continue, 60 countries will not meet the SDG target for under-five mortality. Of these, 47 countries will need to more than double their current rate of progress or reverse an increasing trend and then accelerate progress to achieve the SDG target by 2030. Even more countries are at risk of missing the SDG target for neonatal mortality. A total of 65 countries will need to accelerate the mortality decline to meet the target on time, with 59 of those countries needing to more than double their current rate of decline or reverse an increasing trend to meet the target by 2030. Reaching this target in all countries will avert 8 million under-5 deaths between 2024 and 2030, some 42% of which would be among neonates (3).

2.1.2 Mortality due to noncommunicable diseases

Globally, in 2021, some 18 million people under the age of 70 years died from an NCD, accounting for over half of deaths in that age group. The risk of premature death from NCD, which is measured by SDG indicator 3.4.1, is the unconditional probability of dying from any of the four main NCDs (cardiovascular disease, cancer, diabetes or chronic respiratory disease) between the ages of 30 and 70 years (2). It is estimated that, in 2000, a 30-year-old had a 22.5% chance of dying from one of the four main NCDs before the age of 70 years. This risk fell to 18.4% in 2015 and further to 18.0% in 2019, just prior to the onset of the COVID-19 pandemic (4). The reliability of the estimated risk during 2020–2021 was potentially affected by disruptions associated with the COVID-19 pandemic as countries' ability to monitor trends in NCD mortality was compromised and may have resulted in misclassification of NCD deaths. Where reliable data are available, the risk of premature death from an NCD plateaued during this period.

The observed pre-pandemic trend translates to an ARR of 1.3% between 2000 and 2015 and 0.5% between 2015 and 2019, indicating a remarkable deceleration in progress. With the required ARR to meet the one-third reduction in premature NCD mortality standing at 2.7%, the prepandemic progress since 2015 was equivalent to just one fifth of what is needed to meet the target and calls for coordinated action to prevent and treat NCDs.

The level and pace of progress vary by WHO region. The Eastern Mediterranean Region faced the highest risk of premature NCD mortality in 2000 (26.9%), which progressively declined to 22.7% in 2019, at a level that was only lower than the South-East Asia Region (22.9%) that year. Despite the higher risk compared to most other regions, the Eastern Mediterranean Region is the only region where some acceleration of progress was observed during the SDG era (ARR at 1.2% in 2015–2019 compared with 0.8% in 2000–2015). In contrast, the South-East Asia Region saw overall stagnation in 2000–2019, with a slight uptick in risk between 2015 and 2019. The African Region (23.6% in 2000) experienced a moderate but steady decline, with an ARR at about 0.6% in 2000-2019 and reaching a premature NCD mortality of 21.0% in 2019. The European (23.9% in 2000) and Western Pacific (21.9% in 2000) Regions saw the fastest decline among all regions during both 2000-2015 (ARRs at 2.2% and 1.9%, respectively) and 2015–2019 (ARRs at 1.5% and 1.2%, respectively), reaching premature NCD mortality of 16.3% and 15.7%, respectively, in 2019. The Region of the Americas had the lowest risk of NCD premature mortality in 2000 (18.0%) and remained the lowest in 2019 (13.9%), although the pace of progress slowed, with an ARR at 1.4% in 2000-2015 and 1.1% in 2015-2019.

Based on these results, it is evident that neither the world nor any one region is on track to meet the target of reducing the risk for premature NCD mortality by one third by 2030. The South-East Asia Region, where premature NCD mortality has remained high since 2000, is at greatest risk of missing the target (Fig. 2.3).



Figure 2.3 Observed and projected trends for risk of premature mortality from NCDs (%), by WHO region

Source: WHO (4).

2.1.3 Mortality due to injuries

Road injury

In 2021, an estimated 1.18 million (UI: 1.05–1.32 million) people died due to road injuries globally (SDG indicator 3.6.1). The greatest burden was observed in South-East Asia Region, with around 318 000 (UI: 283 000–358 000) deaths, or 28% of the global burden, and the Western Pacific Region with almost 296 000 (UI: 276 000–316 000) deaths, or 25% of the global burden. Together, these regions account for more than half of the global road traffic deaths. These are followed by the African Region (229 000 [UI: 185 000–274 000] deaths); the Region of the Americas (146 000 (UI: 135 000–158 000) deaths); the Eastern Mediterranean Region (128 000 [UI: 108 000– 145 000] deaths); and the European Region (64 000 [UI: 58 000–68 000] deaths) (*4,5*). The African Region had the highest crude death rate (CDR) in 2021 (19.5 [UI: 15.7–23.4] per 100 000 population). This was nearly three times that of the European Region, which had the lowest rate of 6.8 (UI: 6.2–7.3) per 100 000 population (Fig. 2.4). By World Bank income levels, the highest death rate from road injuries was observed in LICs, at 21.1 (UI: 16.8–25.2) per 100 000 population, while the lowest was in HICs, at 8.1 (UI: 7.5–8.6) per 100 000 population. Disparity between sexes was also evident. Males were at elevated risk of dying from road injuries compared with females. Globally, the male-to-female ratio of CDRs from road injuries was 3.0, ranging from 2.3 in the African Region to above 3.5 in the South-East Asia Region and the Region of the Americas (*4*). Road fatalities declined by 5.5% from 2010-2021, falling from 1.25 million (UI: 1.11–1.39 million) deaths in 2010 (4). This is equivalent to a decline from 17.9 (UI: 15.7-19.9) deaths per 100 000 population to 14.9 (UI: 13.2–16.6) deaths per 100 000 population, and is in spite of the growth in global population and number of vehicles. This indicates that while efforts to enhance road safety are making progress, they remain insufficient to achieve the United Nations Decade of Action for Road Safety 2021-2030 goal of reducing deaths and injuries by half by 2030 (6). There was also a universal decline in the CDRs across all regions in the period. The slowest progress was seen in the Region of the Americas, although rates still declined by 8.9%, from 15.6 (UI: 14.5-16.7) per 100 000 population in 2010 to 14.2 (UI: 13.1-15.4) per 100 000 population in 2021.

Suicide

An estimated 727 000 people died from suicide globally in 2021, corresponding to a crude suicide mortality rates (CDR; SDG indicator 3.4.2) of 9.2 (UI: 6.9–11.6) per 100 000 population. The rate in males (12.4 (UI: 9.6–15.4) per 100 000) was more than double that in females (5.9 (UI: 4.2–7.7) per 100 000). The highest male-to-female ratio was found in the European Region (3.4) and the Region of the Americas (3.7), while the lowest was in the South-East Asia Region (1.4) (4).

Crude suicide mortality rates and their change over time exhibited wide variations across WHO regions. The European Region had the highest rate among all regions in 2000 (21.5 [UI: 19.9-23.0] per 100 000 population), but has seen rapid decline, with the rate falling by over two fifths, to 12.4 (UI: 10.3-14.4) per 100 000 population in 2021. A rapid decline was also seen in the Western Pacific Region, where the rate in 2000 (14.7 [UI: 10.1-16.9] per 100 000 population) fell by 35% to 9.5 (UI: 7.1-12.3) per 100 000 population in 2021. However, a slight increase was seen in the African Region, where the CDR from suicide in 2021 (7.3 [UI: 4.5–10.7] per 100 000 population) was about 5% higher than in 2000 (6.9 [UI: 4.5-10.0] per 100 000). The Region of the Americas has seen a concerning trend, with the rate increasing by nearly 30% from 7.6 (UI: 7.1-8.3) per 100 000 population in 2000 to 9.9 (UI: 8.9-11.0) per 100 000 population in 2021. The South-East Asia Region had the second highest rate after the European Region in 2021 (10.1 [UI: 7.4–12.6] per 100 000), yet experienced a 20.4% decline from the rate in 2000 (12.7 [UI: 9.4-15.1] per 100 000). The Eastern Mediterranean Region continued to have the lowest rate throughout the period, with a 24.3% decline from 4.7 (UI: 2.7–7.4) per 100 000 population in 2000 to 3.6 (UI: 2.0–5.9) per 100 000 population in 2021 (Fig. 2.4) (4).

Homicide

Nearly 484 000 (UI: 360 000–648 000) people were victims of intentional homicide (defined as the unlawful death inflicted upon a person with the intent to cause death or serious injury) in 2021. Despite having a slight increase from 480 000 (UI: 400 000–578 000) in 2000, this corresponded to a decline in CDR of 27.1%, from 7.8 (UI: 6.5–9.4) per 100 000 population (SDG indicator 16.1.1) in 2000 to 6.1 (UI: 4.5–8.2) per 100 000 population in 2021 (Fig. 2.4). About 80% of the victims were men (4).

The WHO Region of the Americas had the highest mortality burden from homicide in 2021, with a total of 198 000 (UI: 169 000-232 000) deaths, equivalent to a CDR of 19.3 (UI: 16.4–22.6) per 100 000 population – triple the global average. This region accounted for 40.9% of the total global homicide deaths but only 12.9% of the global population. In contrast, the Western Pacific Region accounted for about a quarter of the global population, but only 6.1% of global homicide deaths, with a CDR (1.5 (UI: 1.1-2.1) per 100 000 population) that was 8% of that in the Region of the Americas. The African Region had the second highest CDR in homicide mortality in 2021, with a CDR that was only half of that in the Region of the Americas, at 9.7 (UI: 5.9-15.0) per 100 000 population, accounting for 114 000 deaths. Compared with the Region of the Americas and the African Region, the CDR of homicide was also relatively low in the European, South-East Asia and Eastern Mediterranean Regions, at 2.6 (UI: 2.1–3.3) per 100 000 population, 3.7 (UI: 2.7–4.9) per 100 000 population and 5.4 (UI: 3.2-8.5) per 100 000 population, respectively (Fig. 2.4) (4).

Men face disproportionately higher risk of homicide deaths than women and the sex disparity was unevenly distributed across WHO regions. Globally, the male-tofemale ratio for CDR for homicide mortality in 2021 was 4.0, ranging from 2.9 in the European Region to 7.7 in the Region of the Americas. A strong age pattern was also observed, with young adults aged 20–29 years having the highest age-specific mortality rate, peaking at over 12 deaths per 100 000 population globally and up to nearly 40 deaths per 100 000 population in the Region of the Americas (4).

Unintentional poisoning

Globally, about 59 000 (UI: 32 000–91 000) people died from unintentional poisoning in 2021, a decline of approximately 4600 from 2000 (64 000 [UI: 47 000– 92 000]). This represents a decline of a quarter from a CDR of 1.0 (UI: 0.8–1.5) to 0.7 (UI: 0.4–1.1) per 100 000 population (SDG indicator 3.9.3). The highest CDRs in 2021 were observed in the WHO African Region, at 1.2 (UI: 0.7–2.1) per 100 000 population, closely followed by the Western Pacific Region at just under 1.2 (UI: 0.5–1.7)
per 100 000 population. The greatest decline in CDR was achieved in the European Region, with a decline of nearly two thirds between 2000 and 2021, followed by the South-East Asia and Eastern Mediterranean Regions, with a 36% and 30% decline, respectively. However, the rate stagnated in the African Region and a slight increase was seen in the Region of the Americas and the Western Pacific Region (Fig. 2.4) *(4)*.

Sex and age disparities were observed. Men are at higher risk of dying from unintentional poisoning. Men died at a rate 67% higher than women in 2021, primarily due to occupational hazards and higher rates of risky behaviours. The Region of the Americas and the European Region saw the highest male-to-female ratios, at 2.2 and 2.6 respectively, whereas the lowest ratio was in the African Region, at 1.4. Additionally, the youngest and oldest populations were at the highest risk of dying from unintentional poisoning, with those aged under 5 years and 65 years and older accounting for just 18% of the global population but 38% of the global deaths from unintentional poisoning in 2021 *(4)*.



Figure 2.4 Crude death rates for injuries (per 100 000 population), by type, sex and WHO region, 2000 and 2021

AFR: African Region; AMR; Region of the Americas; SEAR: South-East Asia Region; EUR: European Region; EMR: Eastern Mediterranean Region; WPR: Western Pacific Region. Source: WHO (4).

2.1.4 Mortality attributable to environmental risk factors

Air pollution mortality

Almost the entire global population is exposed to outdoor (ambient) air pollution in both cities and rural areas and more than a quarter (2.4 billion) is exposed to dangerous levels of household air pollution. Exposure to air pollution increases the risk for many negative health outcomes (7). Five health outcomes – acute lower respiratory infections, lung cancer, cerebrovascular diseases, ischemic heart disease and chronic obstructive pulmonary disease – are currently included in SDG indicator 3.9.1 on mortality attributed to air pollution from particulate matter. In 2019, an estimated 6.7 million deaths worldwide were attributable to indoor and outdoor air pollution from particulate matter jointly – corresponding to an agestandardized mortality rate of 104 (UI: 81–130) deaths per 100 000 population (*8,9*). People living in LICs and middle-income countries (MICs) disproportionately bear the burden of outdoor air pollution (see also section 2.3.3 subsection "Air pollution").

WASH mortality

Millions of people globally lack access to safe water, sanitation and hygiene (WASH) services and consequently suffer from or are exposed to a multitude of preventable illnesses.

Four health outcomes – diarrhoea, acute respiratory infections, undernutrition and soil-transmitted helminthiases – are included in the SDG indicator 3.9.2 on mortality attributed to unsafe WASH. It is estimated that 1.4 million deaths in 2019 were attributable to unsafe WASH services, of which the vast majority were attributable to diarrhoea (73.9%) and acute respiratory infections (25.4%) (*10*). The global WASH-attributable mortality rate was 18.3 deaths per 100 000 population, ranging from 3.7 in HICs up to 41.7 in LICs (*10*). Among the six WHO regions, the highest mortality rate was found in the African Region (46.7 per 100 000 population) and the South-East Asia Region (29.6 per 100 000 population) in 2019 (*11*). These two regions alone accounted for 79% of the total global deaths due to unsafe WASH (Fig. 2.5).

Figure 2.5 Mortality rate attributed to exposure to unsafe WASH services (per 100 000 population), 2019



Source: WHO (10).

2.2 Infectious diseases

In the infectious disease section, the indicators on new HIV infections, tuberculosis and malaria incidence, hepatitis B and neglected tropical diseases are monitored towards progress within SDG target 3.3. In addition, this section also presents the status of AMR.

2.2.1 HIV

Globally, an estimated 39.9 million (UI: 36.1–44.6 million) people were living with HIV by the end of 2023, of which 30.7 million (UI: 27.0–31.9 million) were receiving lifesaving antiretroviral therapy (*12,13*).

In 2023, an estimated 1.3 million (UI: 1.0–1.7 million) HIV infections occurred, down from 2.8 million (UI: 2.3– 3.6 million) in 2000 and 2.1 million (UI: 1.7–2.7 million) in 2010. Globally, the HIV incidence rate (number of new HIV infections per 1000 uninfected population, SDG indicator 3.3.1) was 0.17 (UI: 0.13–0.21) in 2023, a 48% reduction from 0.32 (UI: 0.25–0.40) in 2010. Regions show different levels of progress since 2010: An 81% increase in HIV incidence rate was observed in the Eastern Mediterranean Region, while it is notable that this region had the lowest number of new infections across all WHO regions in 2023. The South-East Asian and the African Regions recorded a 51% and 69% decrease in HIV incidence rate in the same time period, respectively, while the latter region continued to bear the heaviest HIV burden in 2023. Further disaggregation by sex show that the biggest reduction in HIV incidence rate between 2010 and 2023 was among males in the African Region (-72%) (Fig. 2.6) *(12,14,15)*.

The HIV pandemic and the global response has made progress over the past decade, with fewer people acquiring HIV in 2023 than at any point since the mid-1990s. However, the number of new infections in 2023 was still more than three times the 2025 target of 370 000 or fewer new infections. If this trend continues, the target of ending AIDS as a public health threat by 2030, which requires a 90% reduction in new infections against the 2010 baseline, will not be reached (*16*).





Note: Percentage change in HIV incidence rate was calculated from unrounded incidence rates in the figure and the corresponding text. Source: UNAIDS/WHO estimates, 2024 (14,15).

2.2.2 Tuberculosis

The global increase in new TB cases that has been observed since the start of the COVID-19 pandemic has slowed and is now stabilizing. In 2023, an estimated 10.8 million people (UI: 10.1–11.7 million) developed TB, a slight rise from 10.7 million (UI: 10.0–11.5 million) in 2022. However, this remains higher than the 10.4 million (UI: 9.7–11.1 million) cases in 2021 and 10.1 million (UI: 9.5–10.7 million) in 2020 (*18*).

Among those affected who developed TB in 2023, some 55% were men, 33% were women and 12% were children and young adolescents aged 0–14 years. Most of the

people who develop TB disease each year are in 30 high TB burden countries, accounting for 87% of the global total in 2023 (18).

The global TB incidence rate (SDG indicator 3.3.2) in 2023 was 134 (UI: 125–145) per 100 000 population, similar to the rate in 2022, and an increase from 129 (UI: 121–136) per 100 000 population in 2020. The net reduction between 2015 and 2023 was 8.3%, falling short of the WHO End TB Strategy target of an 80% reduction by 2030 (*17,18*). The African and European regions showed the most progress, with decreases of 24% and 27%, respectively (Fig. 2.7) (*18*).



Figure 2.7 Percentage change in tuberculosis incidence rate between 2015 and 2023, globally and by WHO region

Note: Percentage change in TB incidence rate was calculated from unrounded incidence rates. *Source*: Global tuberculosis report 2024 (18).

2.2.3 Malaria

In 2023, there were an estimated 263 million (UI: 238–294 million) malaria cases globally, an increase of 11 million cases from the previous year. The global incidence rate (SDG indicator 3.3.3) in 2023 was 60.4 (UI: 54.8–67.5) cases per 1000 population at risk, a rise from 58.6 (UI: 53.1–65.6) cases per 1000 population at risk in 2022, and a 4.1% increase from 58.0 (UI: 53.8–62.9) cases per 1000 population at risk in 2015 (Fig. 2.8). The WHO African Region continues to carry the heaviest burden of the disease, accounting for an estimated 94% of cases and 95% of deaths worldwide in 2023 *(19)*. The Global technical strategy for malaria 2016–2030 (GTS) target for reducing malaria case incidence by 90% by 2030 from the 2015 baseline is unlikely to be achieved (20). In 2023, global malaria incidence remained nearly three times higher than the level required to meet the target. However, the WHO South-East Asia Region is on track to meet the GTS 2030 target (19).

Despite these challenges, some notable progress towards elimination has been made. The number of malariaendemic countries declined from 85 in 2022 to 83 in 2023, as two countries maintained zero indigenous cases for three consecutive years. By 2024, a total of 26 countries that were endemic for malaria in 2000 have achieved this milestone (19).





Source: World malaria report 2024 (19).

2.2.4 Hepatitis B

As a major public health challenge of this decade, viral hepatitis B and C lead to chronic disease for hundreds of millions of people and can both cause deaths from liver cirrhosis and liver cancer.

Globally, 1.3 million people died from viral hepatitisrelated causes in 2022, including 1.1 million deaths attributed to hepatitis B alone. This represents an increase of deaths attributed to chronic hepatitis B from the 820 000 in 2019. The Western Pacific Region bears the greatest mortality burden, with 518 000 deaths (or 47.1 of the global total) related to hepatitis B *(21)*. Rapid scale-up of treatment is therefore critical to prevent deaths among people living with chronic hepatitis B. Without this expansion in access, the world will face increasing cases of liver cancer in the next generation, with associated increasing care costs and hepatitis-related deaths.

In 2022, an estimated 254 million people were living with hepatitis B globally. About 5% of the general population in the WHO African Region and Western Pacific Region are living with hepatitis B. Children under the age of five are at risk of vertical (mother-to-child) transmission of hepatitis B, notably in settings with high hepatitis B prevalence. Notably, the African Region is the only region where the hepatitis B prevalence among children younger than five years (SDG indicator 3.3.4) exceeded 1% and yet only 17% of newborns in the Region receive the hepatitis B birth dose vaccination (22). Global new infections of viral hepatitis dropped from 3.0 million in 2019 to 2.2 million in 2022, including 1.2 million newly infected with hepatitis B. The African Region accounted for 63% (or 771 000) of all new hepatitis B infections globally in 2022. This underscores the importance of a focus on scaling up access to viral hepatitis services in Africa, including the hepatitis B birth-dose vaccination, expansion of hepatitis B treatment in the general population and specifically among pregnant women as recently recommended in the 2024 WHO guidelines on people with chronic hepatitis B infection (23).

2.2.5 Neglected tropical diseases

Neglected tropical diseases (NTDs) are a diverse group of conditions with complex epidemiology.¹ They affect some of the world's most vulnerable people, including those living in remote or impoverished communities, creating NTD-related generational cycles of poverty. Addressing the burden of NTDs can break this cycle and

26.0

South-East Asia

Region

contribute to the health, social and economic well-being of entire communities. Globally, in 2023, 1.495 billion people required interventions against NTDs (SDG indicator 3.3.5), about 122 million people fewer than in 2022. The 2023 number represents a 32% decrease from 2010, falling short of the global target of 90% reduction by 2030. The South-East Asia Region and the African Region, home to 41% of the global population, continue to disproportionately bear the highest burden of NTDs, accounting for 87% of the global number of people needing interventions in 2023 (Fig. 2.9) (24,25).

As of December 2024, 54 countries, areas and territories worldwide have eliminated at least one NTD, as acknowledged by WHO. This is still only about halfway towards the global target of at least 100 countries eliminating at least one NTD by 2030 (24,25).

This slow progress reflects the challenges of addressing NTDs on a global scale. Massive efforts, including through innovative solutions, are needed to accelerate progress in the global fight against NTDs, especially in the context of uncertain and evolving health, political and financial landscapes.

12.9

2.1

Region of the

Americas

11.6

0.4

European

Region



10.0

5.2

Eastern

Mediterranean

Region

24.0

5.0

Western Pacific

Region

Figure 2.9 Share of people requiring interventions against NTDs and share of population out of global total, by WHO region, 2023

Region's population requiring NTD intervention as percentage of global population as percentage of global population

15.3

Note: The regions are shown in descending order of the number of people requiring interventions against NTDs. *Source*: Global Health Observatory (24).

African Region

Percentage

20

10

0

¹ NTDs include: Buruli ulcer; Chagas disease; dengue and chikungunya; dracunculiasis; echinococcosis; foodborne trematodiases; human African trypanosomiasis; leishmaniasis; leprosy; lymphatic filariasis; mycetoma, chromoblastomycosis and other deep mycoses; noma; onchocerciasis; rabies; scabies and other ectoparasitoses; schistosomiasis; soil-transmitted helminthiases; snakebite envenoming; taeniasis/cysticercosis; trachoma; and yaws.

2.2.6 Antimicrobial resistance

SDG indicator 3.d.2 monitors the proportion of bloodstream infections caused by antimicrobial-resistant organisms, specifically Escherichia coli resistant to third-generation cephalosporins and methicillinresistant Staphylococcus aureus (MRSA). In 2022, median resistance proportions were 45.5% (interquartile range [IQR]: 18.2–70.6) for E. coli and 32.2% (IQR: 12.5–54.6) for MRSA, up from 26.9% (IQR: 11.1–42.4) and 13.4% (IQR: 3.6–36.7) in 2016 (*27*).

These figures should be interpreted with caution due to changes in countries reporting data, with progressively more data collected from resource-limited settings over time. In 2016, only 16 countries reported data on E. coli resistance and 15 on MRSA, compared to 85 and 86 countries, respectively, in 2022 (Fig. 2.10). Many LICs and MICs have limited testing coverage, often restricted to tertiary hospitals and research facilities, where patients are more likely to have received multiple antibiotics, leading to higher and potentially biased resistance rates. Countries with better testing coverage reported much lower resistance proportions in 2022: 11.4% (IQR: 9.4– 16.4) for E. coli and 8.5% (IQR: 3.3–27.9) for MRSA, based on median resistance from 21 and 22 countries with testing coverage above the 75th percentile, respectively (27). The 2024 WHO Bacterial Priority Pathogen List (26) classifies E. coli resistant to third-generation cephalosporins as a critical priority due to increasing trends, while MRSA is considered as high priority based on its prevalence and impact in health-care settings, and decreasing trends.

Current global AMR data reflect the challenges of earlystage surveillance systems. Strengthening these systems, particularly in LICs and MICs, is essential for improving data accuracy and representativeness to inform interventions to prevent and reduce AMR. The Global Antimicrobial Resistance and Use Surveillance System (GLASS) dashboard (27) provides more information on the current AMR situation in the human health sector.





Source: Data from GLASS dashboard (27).

2.3 Risk factors for health

A risk factor for health is any attribute, characteristic or exposure of an individual that increases their likelihood of developing a disease or injury. This section presents progress towards achieving the SDG targets related to nutritional, behavioural and environmental risk factors that may affect individual and population health.

2.3.1 Nutritional risk factors

Multisectoral factors affect the nutritional status of individuals and populations. Mitigating the risks of malnutrition – manifested as undernutrition (wasting, stunting, underweight), micronutrient-related conditions, overweight and obesity – requires multifaceted approaches through nutrition-specific (e.g. food, diet) and nutrition-sensitive strategies (e.g. WASH, education, gender). In fact, undernutrition and food safety are inextricably linked and comprise the essential components of a healthy and sustainable food system. The following sections present the latest available estimates for the global SDG indicators for stunting, wasting and overweight among children under 5 years of age, and anaemia in women aged 15–49 years.

Child malnutrition

The indicators to monitor progress in reducing child malnutrition among children under 5 years of age include the prevalence of stunting (SDG indicator 2.2.1), and the prevalence of wasting and overweight (SDG indicator 2.2.2).

The global prevalence of stunting among children under 5 years of age had been steadily declining from 33.1% (UI: 32.7–33.5%) in 2000 to 22.4% (UI: 22.0–22.8%) in 2020 but had slightly increased since to reach 23.2% (UI: 22.6–23.8%) in 2024 (Fig. 2.11) *(28)*.

The number of children affected by stunting worldwide fell from 180 million in 2012 to 150 million in 2024 (Fig. 2.11) – a net reduction of 17% over the period (28). This pace of reduction is insufficient to achieve the global target of halving the number of children affected by stunting by 2030 from the 2012 baseline (28,29).

Similarly, the world is off track to reach the target of reducing the prevalence of wasting and overweight to less than 3% each by 2030 (Fig. 2.12) *(29)*. In 2024, the global prevalence of wasting among children under 5 years of age stood at 6.6% (UI: 5.3–8.1%), having declined by 11% since 2012, when the prevalence was 7.4% (UI: 6.3–8.7%) *(28)*.

While there was a sign of slight decline in the global prevalence of overweight among children under 5 years of age in the 2010s, it has slightly increased again since 2020 to reach 5.5% (UI: 4.9–6.1%) in 2024. Between 2012 and 2024, there was a net increase of 3% in the global prevalence of overweight, necessitating a reversal of trend and an acceleration of progress to meet the 2030 target *(28)*.



Figure 2.11 Global prevalence of stunting among children under 5 years of age, 2000–2024, and the 2030 target

Source: UNICEF/WHO/World Bank Group joint child malnutrition estimates, 2025 (28).





Source: UNICEF/WHO/World Bank Group joint child malnutrition estimates, 2025 (28).

Anaemia among women

Anaemia is an important global health issue mainly affecting young children, menstruating girls and women, and pregnant and postpartum women. It can affect school performance and productivity in adult life leading to economic losses. During pregnancy, anaemia has been associated with poor maternal and birth outcomes.

The global prevalence of anaemia among all women 15–49 years of age (SDG indicator 2.2.3) declined marginally from 29.3% (UI: 26.4–32.4%) in 2000 to 27.6% (UI: 25.3–29.7%) in 2012 but slightly increased since to reach 30.7% (UI: 26.6–34.9%) in 2023 (Fig. 2.13). At this rate, the world is off track to meet the target of 50% reduction by 2030 (from a 2012 baseline) *(29,30)*.

Among pregnant women globally, the prevalence decreased from 40.3% (UI: 37.7–43.2%) in 2000 to 35.3% (UI: 33.3–37.3%) in 2012, however, it remained stagnant since. In 2023, the prevalence was 35.5% (UI: 32.1– 39.1%). By contrast, prevalence among non-pregnant women only decreased slightly from 28.7% (UI: 25.7– 31.9%) in 2000 to 27.1% (UI: 24.9–29.4%) in 2012 and increased to 30.5% (UI: 26.2–34.8%) in 2023.

Regional trends in the prevalence of anaemia among all women 15–49 years of age indicate a slight increase in all WHO regions since 2012 except in the Western Pacific Region, where it has remained unchanged (Fig. 2.13). The South-East Asia Region shows the highest level of anaemia since 2000. In 2023, anaemia affected almost half (46.4%, UI: 35.5–56.1%) of all women aged 15–49 years in the region.

2.3.2 Behavioural risk factors

Among the established risk factors associated with NCDs are modifiable behaviours such as tobacco use, alcohol consumption, physical inactivity and unhealthy diet. This section summarizes the global progress towards the reduction of tobacco use and alcohol consumption.

Tobacco use

Tobacco use is a major risk factor for cardiovascular and respiratory diseases, different types of cancer and many other debilitating health conditions. There is no safe level of exposure to tobacco, and all forms of tobacco are harmful.

In 2022, an estimated 1.25 billion persons aged 15 years and older globally were current tobacco users. This corresponds to an age-standardized prevalence (SDG indicator 3.a.1) of 20.9% (UI: 18.8–22.9%), a 21% relative decline from 26.4% in 2010. If this pace of decline is maintained, the global prevalence would be reduced by 25% by 2025, short of the 30% reduction target. Nevertheless, the African Region and the South-East Asia Region are currently on track to achieve the target, and other regions have also achieved reductions in average prevalence (*31,32*).





Source: WHO (30).

Prevalence remained much higher among men than women globally. In 2022, there were four times as many male tobacco users as female users among those aged 15 years and older. The prevalence among female tobacco users fell by 34% between 2010 and 2022, exceeding even the 2025 global target. Among male users, prevalence has so far declined by only 17%. However, the fastest reduction among male tobacco users was seen in the "initiation ages" (age group 15–24 years), a 20% relative reductions in prevalence of tobacco use among men in the future (*31*).

Alcohol consumption

Drinking alcohol is associated with risks of developing NCDs such as liver diseases, cardiovascular diseases and different types of cancers, as well as mental health and behavioural conditions such as depression, anxiety and alcohol use disorders. Alcohol consumption also causes significant harm to others, as it can lead to injuries such as those caused by road traffic crashes. The Global alcohol action plan 2022–2030 includes a target of at least a 20% relative reduction in the harmful use of alcohol by 2030 (in comparison with 2010) (33). SDG indicator 3.5.2 is defined as total alcohol per capita consumption (APC) in persons aged 15 years and older within a calendar year, measured in litres of pure alcohol. Globally, the total APC declined by 12% from 5.7 litres (UI: 5.4–6.2) in 2010 to 5.0 litres (UI: 4.7–5.4) in 2022. If this rate of reduction is maintained, the global total APC will reach 4.6 litres in 2030, reaching the 20% reduction target. This is remarkable progress, especially given that in the early 2010s the global total APC was increasing, continuing the upward trend since 2000s, driven mainly by the rise in total APC in the South-East Asia Region and the Western Pacific Region (Fig. 2.14) (*34*).

The South-East Asia Region has curbed the rise and the APC level in 2022 had returned to its 2010 level. The Western Pacific Region has reversed the trend and is now on track to reach 20% reduction in APC by 2030. The Eastern Mediterranean Region has remained the region with the lowest APC across all regions. The European Region continues to have the highest total APC, although it has declined by 12% since 2010 (Fig. 2.14) (34).





Source: WHO (34).

2.3.3 Environmental risk factors

Water, sanitation and hygiene

Safe drinking water, sanitation and hygiene (WASH) is crucial to human health and well-being. Contaminated water and poor sanitation are linked to transmission of diseases such as cholera, diarrhoea, dysentery, hepatitis A, typhoid and polio.

The global coverage of safely managed drinking water (SDG indicator 6.1.1) increased from 69% in 2015 to 73%

in 2022. During the same period, the global coverage of safely managed sanitation (SDG indicator 6.2.1) rose from 49% to 57% and that of basic hygiene services (SDG indicator 6.2.1) increased from 67% to 75%. Despite this progress, if current trends continue the world will not achieve universal coverage of safe WASH by 2030 (*35*).

Globally, and in all WHO regions with aggregate data for 2022, coverage of safe WASH is higher in urban areas than in rural areas, except in the South-East Asia Region, where the coverage of both safely managed drinking water and safely managed sanitation was higher in rural areas (Fig. 2.15) *(35)*.

Figure 2.15 Coverage of safe WASH by location, globally and by WHO region, 2022



Note: Absence of bar for some services and locations means no aggregate statistics are available for 2022. *Source*: UNICEF and WHO (*35*).

SDG indicator 6.3.1 monitors the proportion of total, domestic and industrial wastewater flows that are safely treated. Many countries, however, are at an early stage of developing wastewater monitoring systems and there is currently a lack of reporting, especially from industrial sources. Among the 73 countries (representing 42% of the global population) reporting volumes of total wastewater generated and treated, 76% of total wastewater flows received some level of treatment in 2022. Among the 42 countries (representing 12% of global population) reporting the type of treatment, 60% of total wastewater flows was "safely" treated (i.e. had at least secondary treatment, or treatment complying with relevant standards) (36). An estimated 56% of household wastewater was safely treated in 2024, based on data from 142 countries and areas (representing 88% of all household wastewater flows and 84% of the global population) (37). While current data are insufficient to accurately assess progress since the start of the SDG era (mainly due to the lack of baseline data for 2015), data availability has greatly improved. Still, there is little evidence of improvements in the proportion of wastewater that is safely treated globally since global reporting began in 2021.

Official development assistance (ODA) disbursements to the water sector (SDG indicator 6.a.1) declined slightly from US\$ 8.9 billion in 2022 to US\$ 8.7 billion in 2023,

however they remained higher than disbursements in 2020 and 2021, which had the lowest levels since the start of the SDGs in 2015. ODA commitments to the water sector decreased more substantially, from US\$ 11.4 billion in 2022 to US\$ 10.2 billion in 2023 (*38*).

Air pollution

Air pollution, both indoor and outdoor (ambient), poses major risks to health. The main pathway of exposure to air pollution is through the respiratory tract. However, the effects on health extend beyond respiratory diseases, as some air pollutants can penetrate the bloodstream via the lungs and circulate throughout the entire body leading to systemic inflammation and carcinogenicity.

Simple stoves paired with charcoal, coal, crop waste, dung, kerosene or wood are an important source of household air pollution. The latest estimates reveal that 2.1 billion (UI: 1.9–2.5 billion) people worldwide were still relying on these polluting cooking fuels and technologies in 2023 – equivalent to slightly over a quarter (26%, UI: 23–30%) of the global population. The proportion of the global population with primary reliance on clean cooking fuels and technologies had increased from 49% (UI: 45–53%) in 2000 to 64% (UI: 60–68%) in 2015 and 74% (UI: 70–77%) in 2023 (Fig. 2.16) *(39)*. Under current trends, the world will miss the target of universal access by 2030.



Figure 2.16 Global trends in the access to clean cooking fuels and technologies, 2000–2023

In terms of ambient air pollution, SDG indicator 11.6.2 tracks the annual average concentration of fine particulate matter in urban areas. Globally, air quality levels for particulate matter are still more than six times the recommended level for protecting public health according to the WHO air quality guidelines, which is set at the annual mean of $5 \mu g/m^3$. Nearly all populations in HICs, LICs and MICs are exposed to unsafe levels of air pollution. However, the proportion of population in LICs and MICs exposed to levels above the interim target 4 (annual mean of $10 \mu g/m^3$) is twice that in HICs, at 93.5% and 47.1%, respectively. The proportion of population in LICs and MICs exposed to levels exceeding interim target 1 (annual mean of $35 \mu q/m^3$) is over 10 times that in HICs (42.0% and 3.7%, respectively). In addition, people living in LICs and MICs are less likely to have air quality standards that can protect them. The recent publication of the WHO air quality standard database shows that only 32% of LICs and MICs (17 countries) have annual fine particulate matter (PM2.5) standards while 76% of HICs (44 countries) have such protective standards (40).

2.3.4 Risks to girls' and women's health

Adolescent pregnancy

Adolescent pregnancy is a worldwide issue with significant health, social and economic impacts. Teenage

mothers aged 10–19 years face higher risks of eclampsia, puerperal endometritis and systemic infections than women aged 20–24 years, and babies of teenage mothers face higher risks of low birth weight, preterm birth and severe neonatal conditions.

The global adolescent birth rate (ABR; SDG indicator 3.7.2) among girls aged 15–19 years declined by 17%, from 45.9 births per 1000 girls in 2015 to 38.3 births per 1000 girls in 2024. The global ABR among girls aged 10–14 years fell at almost double the rate, with a 31% decline from 1.5 births per 1000 girls in 2015 to 1.0 birth per 1000 girls in 2024. While this decline is encouraging, there were still far too many births among adolescent mothers in 2024, accounting for about 9.4% of all births globally (*41*).

Progress is uneven among regions (Fig. 2.17), as well as among and within countries. The African Region continued to have the highest ABR in both adolescent age groups (10–14 years and 15–19 years). In 2024, the ABR among girls aged 15–19 years was 90.5 births per 1000 girls, over eight times that of the region with the lowest ABR, the Western Pacific Region. The gaps in ABR among girls aged 10–14 years were even larger: at 3.0 births per 1000 girls, it was 35 times that of the regions with the lowest ABR, the European Region. By region and age group, the biggest relative reduction in ABR occurred in the South-East Asia Region, where ABR among girls aged 10–14 years fell by 61%, from 0.8 births per 1000 girls in 2015 to 0.3 births per 1000 girls in 2024 (*41*).



Figure 2.17 Adolescent birth rate, globally and by WHO region, 2015–2024

AFR: African Region; AMR; Region of the Americas; EMR: Eastern Mediterranean Region; EUR: European region; SEAR: South-East Asia Region; WPR: Western Pacific Region.

Note: The regions are shown in descending order of the adolescent birth rate (ABR) in 2024. Values shown above the 2024 bars denote the relative reduction of ABRs between 2015 and 2024.

Source: World population prospects 2024 (41).

Violence against women

Violence against women and girls – particularly intimate partner violence (IPV) and sexual violence – is a violation of human rights and a major public health problem. Such violence often has devastating impacts on the lives and well-being of the victims, as well as their children and families.

According to the latest available global estimates, in 2018 over a quarter (26%, UI: 22–30%) of ever-partnered women aged 15 years or older had experienced IPV at least once in their lifetime (SDG indicator 5.2.1) and 10% (UI: 8–12%) in the past 12 months (42). There has been insufficient progress in reducing IPV against women and girls since 2000 (43). In 2018, an estimated 6% (UI: 4–9%) of all women globally had been subjected to non-partner sexual violence in their lifetime since the age of 15 (SDG indicator 5.2.2). The true prevalence is likely to be much higher (42).

Population-based surveys which record survivors' responses provide the best information to measure prevalence of IPV and sexual violence. However, even such surveys often have limitations, such as variations in case definitions and recall periods, stigma and repercussions associated with the disclosure of violence against women, and limited data collection from women aged 50 years and older.

2.4 Universal health coverage and health systems

Universal health coverage aims to ensure that all individuals and communities have access to the quality health services they need without suffering financial hardship. When more people have access to affordable quality health services, it leads to better overall public health and living standards, including, but not limited to, improved life expectancy, lower rates of preventable diseases and lower rates of poverty due to medical costs.

2.4.1 Service delivery

To achieve UHC and improve health outcomes, health systems need strong service delivery mechanisms that ensure comprehensive service coverage, a capable health workforce and adherence to international health regulations (IHRs) to address both routine and emergency health needs.

Service coverage

While gains in UHC service coverage index (UHC SCI; SDG indicator 3.8.1) have been observed globally over the past

two decades, progress has slowed in recent years (44). The UHC SCI is composed of 14 tracer indicators selected as a manageable subset to represent overall coverage. These indicators are organized into four domains: reproductive, maternal, newborn and child health (RMNCH); infectious diseases; NCDs; and service capacity and access.

The progress made in the UHC SCI has not been even across domains (44). Figure 2.18 shows the change since 2000 in UHC SCI and selected tracers or related health SDG service coverage indicators.² Globally, service coverage tracer indicators improved by a range of 3–75 percentage points from 2000 to the latest available year (2019–2024). There remains significant scope for progress in the area of NCDs, which contribute to the largest proportion of disease burden globally. Hypertension and diabetes treatment coverage (age-standardized) remains below 50% of the target population globally (42% and 40% in 2019 and 2022, respectively) and progress needs to be accelerated (14,15,18,31,35,44–48).

Service coverage indicators (scale of 0–100) with a global baseline of 2000 and a latest available of 2019 or later.

Figure 2.18 Universal health coverage service coverage index (UHC SCI) and selected tracers globally, 2000 and latest available year (2019–2024)



ART: Antiretroviral therapy; DTP3: diphtheria, tetanus toxoid and pertussis vaccination; TB: tuberculosis. ^a Diabetes treatment will be a UHC SCI tracer in the next round of reporting, see Box 2.2 for more information.

Source: Sanitation (35), UHC SCI (44), ART (14,15), TB (18), hypertension (45), diabetes (46), tobacco (31), family planning (47), DTP3 (48).

In the domain of RMNCH, the proportion of women of reproductive age having their need for family planning met with modern methods (SDG indicator 3.7.1) has seen a modest improvement of an average of 1.2 percentage points in the each of the last two decades globally, with no WHO region reaching universal coverage (range 58.1–87.3%) in 2024 (47). The global proportion of births attended by skilled health personnel (SDG indicator 3.1.2) increased by 26 percentage points between 2000 and 2024, to 87%. As of 2024, the WHO Region of the Americas and Western Pacific and European Regions have achieved universal or nearly universal coverage, while only 74% of births are attended by a medical doctor, nurse or midwife in the WHO African Region (49).

Vaccination rates of all four childhood vaccines tracked within SDG indicator 3.b.1³ stagnated or declined between 2020 and 2022. For third dose diphtheriapertussis-tetanus containing vaccine (DTP3), which is a useful marker of vaccination system functioning, in 2023 vaccination rates had still not fully returned to their 2019 pre-pandemic level, suggesting substantial residual dysfunction in global vaccine delivery systems. Vaccination rates for the other three vaccines increased, although this was partially due to new introductions. Globally, DTP3 still has the highest coverage of the four vaccines, at 84% in 2023.

The WHO Region of the Americas has the highest coverage of the human papillomavirus (HPV) vaccination among girls aged 9-14 years, reaching 55% within 18 years of introduction of the vaccine. However, globally HPV coverage remains low, meaning millions of girls will remain susceptible to cervical cancer over their lifetimes. Among WHO regions, the European Region has the highest uptake of third dose pneumococcal conjugate vaccine (PCV3), with coverage at 86% in 2023. In 2023, the WHO European and Western Pacific Regions had DTP3 and measles-containing-vaccine second-dose (MCV2) coverage of 90% or higher, and the South-East Asia Region reached 90% for DTP3 (48). However, the 90% global target set by the Immunization Agenda is unlikely to be met for any of these vaccinations if current trends continue (50).

Data collected from about 20 countries between 2015 and 2018 on treatment for alcohol and drug use disorders (SDG indicator 3.5.1) indicate a median contact coverage of 3.2% and 10.0%, respectively. About 40% of the 145 countries who responded to Global Survey 2019 on Progress on SDG health target 3.5 do not collect any data

³ Diphtheria-tetanus-pertussis (DTP3), measles-containing-vaccine second-dose (MCV2), pneumococcal conjugate 3rd dose (PCV3) and human papillomavirus (HPV) vaccine.

on service provision and service utilization for treatment of alcohol and drug use disorders (*51*). Alcohol and drug use are risk factors for multiple health conditions, such as cardiovascular diseases, liver diseases, cancer, respiratory diseases, injuries and mental health disorders, all of which contribute to NCD-related morbidity and mortality (*52*). Surveys are essential for gathering population health data, including, but not limited to, fertility rates, nutrition, child health and disease prevalence. The data collected by surveys support informing health policy, improving health systems and identifying emerging health issues (see Box 2.1).

Box 2.1 Demographic and Health Surveys in universal health coverage service coverage index (UHC SCI)

Over half of the UHC SCI tracers use surveys such as Demographic and Health Surveys (DHS) as a data source. The DHS programme was initiated in the early 1980s by the United States Agency for International Development (USAID) with the aim of providing countries with reliable and timely data on critical health and demographic indicators. It aimed to fill a gap in data collection, particularly in developing countries where such information was often lacking or outdated. The first survey was conducted in El Salvador in 1985. Since 2000, there have been around 300 DHS conducted (see Fig. 2.19) *(53)*. The suspension of USAID funding delays crucial public health data, resulting in the short-term in an evidence gap to allocate resources and develop policies. Without immediate actions to replace this data collection, the evidence gap will lead to less effective health interventions and widening health disparities. As statistical offices and ministries of health develop a strategy to mitigate the loss of the DHS, other questionnaires such as the WHO World Health Survey Plus *(54)* can guide comprehensive data collection to monitor progress towards population health targets and health-related SDGs.



Figure 2.19 Number of demographic and health surveys conducted by DHS programme by country, 2000–2024

Note: Includes all survey types with data available as of 27 March 2025. *Source*: DHS (*53*).

Health workforce

Health workers are crucial for achieving UHC. Their role goes beyond providing medical care: an ample, welltrained and properly distributed workforce ensures that health services are accessible, effective and of good quality for all, regardless of income or place of residence.

The estimated global stock of health workers exceeds 70 million in 2023, with the five occupations included in SDG target 3.c (dentists, medical doctors, midwifery personnel, nursing personnel and pharmacists) accounting for 52 million (55). Data for 2023 show a 26% increase in the density of dentists, medical doctors, midwifery personnel, nursing personnel and pharmacists since 2013, but with significant differences by region and income group (for instance, on average, 1 for every 64 people in HICs versus 1 for every 621 people in LICs) (56). The estimated global shortage of 15.4 million health workers in 2020 decreased to 14.7 million in 2023. The projected 2030 shortage of 11.1 million shows slow progress in closing the gap (Fig. 2.20), with the WHO African and Eastern Mediterranean Regions projected to bear nearly 70% of the shortage. HICs and uppermiddle-income countries (UMICs), which account for 53% of the global population and 80% of workers in the five occupations under SDG target 3.c, have the highest densities of health workers. Nevertheless, these countries will likely need to recruit 8.4 million additional health workers by 2030 to maintain density levels and meet the rising demands of growing and ageing populations, potentially accelerating international migration of health workers.

Figure 2.20 Distribution of the global health workforce shortage by WHO region (for 2013, 2020 and 2023, and projected for 2030)



Note: Percentages represent the relative share of the global health workforce shortage. *Source*: NHWA (56).

International health regulations

IHRs are essential to service delivery as they ensure countries have the necessary capacities within their health systems to prevent, prepare, detect in a timely manner and respond to public health emergencies. IHRs therefore contribute to global health resilience and the achievement of sustainable health outcomes for all populations.

IHRs States Parties self-assessment annual reporting tool (SPAR) second edition (*57*), which assesses 15 capacities,

was launched in 2021 and is in its fourth year of reporting. Globally, the capacities average (SDG target 3.d.1) has wavered around 64% between 2021 and 2024. The three main achievement and challenges capacities measured by SPAR (Fig. 2.21) have changed minimally (1 point) globally, with the exception of challenge C1 (Policy, legal and normative instruments to implement IHR), which improved by 3 points globally and by 11 in the South-East Asia Region. The WHO African Region improved its score (from 2 to 6 points) in all three main challenges and two of the achievements between 2021 and 2024 (*58*).

Figure 2.21 Global top three challenges and achievements scores in 2024, by WHO region

Top global challenges	Global	African Region	Region of the Americans	South-East Asia Region	European Region	Eastern Mediterranean Region	Western Pacific Region
C14 Chemical events	53	35	51	53	69	55	53
C1 Policy, legal and normative instruments to implement IHR	55	38	55	61	64	61	62
C15 Radiation emergencies	57	38	53	55	77	67	47
Top global achievements							
C5 Surveillance	80	73	79	83	85	80	79
C7 Health emergency management	71	60	74	75	75		76
C8 Health services provision	71	56	73	78	79	75	73

Scores: dark green: > 80; light green: 61–80; orange: 41–60; yellow: 21–40. Change from 2021: bold: increased by more than 1 point; italic: decreased by more than 1 point; not bold or italic: change is 1 point or less. Source: eSPAR (58).

2.4.2 Health financing

Sustainable, efficient and effective health financing is critical for achieving UHC. By pooling resources, reducing out-of-pocket payments and ensuring financial protection, health financing allows countries to provide equitable access to quality health care for all populations.

Financial protection

The financial protection dimension of UHC is achieved when there are no financial barriers to accessing needed health services and goods, and out-of-pocket (OOP) health spending is not a source of financial hardship. SDG indicator 3.8.2 measures the proportion of people experiencing catastrophic OOP health spending. The latest data show that the global proportion of the population spending more than 10% of their household budget on health OOP increased from 2000 to 2019, reaching 13.5%, while the global proportion of the population spending more than 25% reached 3.8%. It is also important to track how many people are impoverished or further impoverished by OOP health spending, as even relatively small OOP payments can threaten the living standards of people living near or in poverty. In 2019, some 344 million people were pushed or further pushed into extreme poverty by OOP health spending, while 13.5% of the global population spent more than 10% of their household budget on OOP payments for health. The proportion of people facing impoverishing OOP spending at the extreme poverty line declined from 22.2% in 2000 to 4.4% in 2019. In contrast, the share at the relative poverty line rose from 11.8% in 2000 to 16.7% in 2019 (Fig. 2.22). These trends suggest that while OOP health spending was less likely to push people into extreme poverty by 2019, it still disproportionately affected those in the lower income brackets (44).





Source: WHO and World Bank (44).

Domestic and international funding

Domestic public funds provide long-term sustainability, equity and a foundation for health systems, while international funds address funding gaps, support global initiatives and bring innovation to health care delivery.

The increase in the prioritization of health within general government spending seen in 2020 and 2021 (the health component of SDG indicator 1.a.2) among middleand high-income groups was not sustained in 2022, and even declined slightly in LICs, LMICs and UMICs. However, the share of health expenditure within total government spending remained higher in 2022 than in 2019 in all income groups (*59*). Government funding is essential to subsidize health services and ensure they are affordable for everyone, including low-income or vulnerable populations.

Development assistance to medical research is vital for addressing both emerging health challenges and

ongoing diseases, particularly in LICs and MICs, where the basic health portion addresses fundamental health-care needs that are critical to UHC. In 2023, LICs received the highest weighted average ODA per capita (SDG indicator 3.b.2), at US\$ 6.04, compared with US\$ 0.88 and US\$ 0.43 for the LMICs and UMICs, respectively. However, ODA varies significantly by country; the five countries with the highest per capita ODA are mostly island states in the middle- and high-income groupings (60).

Since the global indicator framework for SDGs was agreed in 2017 *(61)*, challenges including the rising burden of NCDs, the impact of global health crises like the COVID-19 pandemic, workforce shortages and financial sustainability, have increased. Indicators used to monitor UHC need to adapt to the changing world, improvements in data availability and understanding of populations being left behind (Box 2.2).

Box 2.2 Revising SDGs to improve monitoring of progress towards UHC

Together, SDG indicators 3.8.1 and 3.8.2 are essential to monitoring progress towards UHC: 3.8.1 aims to assess whether needed services are available and used by those in need, while 3.8.2 assesses whether people are financially burdened by the cost of care. The two indicators were revised as part of the Inter-Agency and Expert Group on Sustainable Development Goal Indicators 2025 comprehensive review *(62)* and approved by United Nations Statistical Commission in March 2025 *(63)*.

The revision of SDG indicator 3.8.1 follows the same principals as its original construction (64) but aims to update to indicators with improved data availability for better tracking and alignment with existing SDG indicators. The tracers were selected as a subset of overall coverage and all equally contribute to the UHC SCI. The revision will apply weights based on the population affected by each tracer to better reflect the service burden on a health system; the four domain areas will still equally contribute to the overall SCI.

The revision of SDG indicator 3.8.2 captures financial hardship caused by both substantial and impoverishing OOP health spending. Initially, the budget share definition was retained for SDG indicator 3.8.2, with a 10% threshold used to identify large OOP spending. This was intended to identify hardship in poorer populations. However, global monitoring reports on UHC have shown that many people living in poverty, who spend less than 10% of their household budget on health, are still pushed or further pushed into extreme or relative poverty. The revised indicator is defined as the proportion of the population with OOP household expenditure on health exceeding 40% of the household discretionary budget. A household's discretionary budget is defined as household total consumption expenditure or income net of the societal poverty line (SPL). The SPL is used to assess the cost of basic needs across countries at all income levels. The discretionary budget approach takes into account that poorer households allocate most of their resources to necessities and in effect can spend very little on health OOP without struggling financially, while wealthier households can spend much more before OOP spending leads to financial hardship.

These revisions align with the SDG principle of ensuring no one is left behind and aim to better reflect the realities of service coverage and financial hardship across countries. The new definitions will be reflected in the next *Tracking universal health coverage: global monitoring report (65)* to be published at the end of 2025.

2.5 SDG progress to date

As the world moves closer to 2030, progress is insufficient to meet the health-related SDG and other global targets. While there have been improvements for most of the indicators reviewed in this chapter, only one – reducing alcohol consumption – is on track to meet the 2030 target at the global level. The assessment of progress in selected health-related SDG indicators by WHO region is presented in Annex 1. Concerted efforts are needed to accelerate progress and reverse regressing trends globally, as well as at regional, country and community levels.

References

- 1. Trends in maternal mortality estimates 2000 to 2023: estimates by WHO, UNICEF, UNFPA, World Bank Group and UNDESA/Population Division. Geneva: World Health Organization; 2025. Licence CC BY-NC-SA 3.0 IGO.
- Work of the Statistical Commission pertaining to the 2030 Agenda for Sustainable Development: resolution/ adopted by the General Assembly. New York: United Nations; 2017 (A/RES/71/313; https://digitallibrary.un.org/ record/1291226?v=pdf).
- United Nations Inter-agency Group for Child Mortality Estimation (UN IGME). Levels & trends in child mortality: report 2024 – estimates developed by the United Nations Inter-agency Group for Child Mortality Estimation. New York: United Nations Children's Fund; 2025.
- 4. Global health estimates 2021. Geneva: World Health Organization; 2024 (https://www.who.int/data/global-healthestimates, accessed on 19 February 2025).
- Global status report on road safety 2023. Geneva: World Health Organization; 2023 (https://iris.who.int/ handle/10665/375016). Licence: CC BY-NC-SA 3.0 IGO.
- Improving global road safety: resolution/adopted by the General Assembly. New York: United Nations; 2020 (A/RES/74/299; https://digitallibrary.un.org/ record/3879711?ln=zh_CN&v=pdf).
- Air pollution: impact. World Health Organization; 2025 (https://www.who.int/health-topics/air-pollution#tab=tab_2, accessed on 2 March 2025).
- 8. Ambient (outdoor) air pollution [fact sheet]. World Health Organization; 24 October 2024 (https://www.who.int/newsroom/fact-sheets/detail/ambient-(outdoor)-air-quality-andhealth).
- 9. Household air pollution [fact sheet]. World Health Organization; 16 October 2024 (https://www.who.int/newsroom/fact-sheets/detail/household-air-pollution-and-health).
- Burden of disease attributable to unsafe drinking-water, sanitation and hygiene, 2019 update. Geneva: World Health Organization; 2023. Licence: CC BY-NC-SA 3.0 IGO.
- Mortality rate attributed to exposure to unsafe WASH services (per 100 000 population) (SDG 3.9.2). In: The Global Health Observatory [website]. World Health Organization; (https:// www.who.int/data/gho/data/indicators/indicator-details/ GHO/mortality-rate-attributed-to-exposure-to-unsafe-washservices-(per-100-000-population)-(sdg-3-9-2), accessed on 2 March 2025).
- 12. HIV statistics, globally and by WHO region, 2024 [fact sheet]. Geneva: World Health Organization; 2024 (https://www.who. int/teams/global-hiv-hepatitis-and-stis-programmes/hiv/ strategic-information/hiv-data-and-statistics).
- 13. The urgency of now: AIDS at a crossroads. 2024 global AIDS update. Geneva: Joint United Nations Programme on HIV/ AIDS; 2024 (https://www.unaids.org/sites/default/files/ media_asset/2024-unaids-global-aids-update_en.pdf).
- 14. HIV. The global health observatory [online database]. Geneva: World Health Organization; (https://www.who.int/data/gho/ data/themes/hiv-aids, accessed on 26 March 2025).
- 15. AIDSinfo [online database]. Geneva: Joint United Nations Programme on HIV/AIDS (UNAIDS); (http://aidsinfo.unaids. org, accessed on 26 March 2025).

- Global AIDS strategy 2021–2026. End inequalities, end AIDS. Geneva: Joint United Nations Programme on HIV/AIDS; 2021 (https://www.unaids.org/sites/default/files/media_asset/ global-AIDS-strategy-2021-2026_en.pdf).
- 17. The end TB strategy. Geneva: World Health Organization; 2015 (WHO/HTM/TB/2015.19; https://iris.who.int/ handle/10665/331326).
- Global tuberculosis report 2024. Geneva: World Health Organization; 2024 (https://iris.who.int/ handle/10665/379339). Licence: CC BY-NC-SA 3.0 IGO.
- World malaria report 2024: addressing inequity in the global malaria response. Geneva: World Health Organization; 2024 (https://iris.who.int/handle/10665/379751). Licence: CC BY-NC-SA 3.0 IGO.
- 20. Global technical strategy for malaria 2016–2030, 2021 update. Geneva: World Health Organization; 2021 (https://iris.who.int/ handle/10665/342995).
- 21. Global hepatitis report 2024: action for access in low- and middle-income countries. Geneva: World Health Organization; 2024. Licence: CC BY-NC-SA 3.0 IGO.
- 22. WHO. Hepatitis B vaccination coverage [database]. World Health Organization; 2022 (https://immunizationdata. who.int/global/wiise-detail-page/hepatitis-bvaccination-coverage?CODE=Global&GROUP=WHO_ REGIONS&ANTIGEN=HEPB_BD&YEAR=, accessed on 2 March 2025).
- 23. Guidelines for the prevention, diagnosis, care and treatment for people with chronic hepatitis B infection. Geneva: World Health Organization; 2024. Licence: CC BY-NC-SA 3.0 IGO.
- 24. Neglected tropical diseases. The global health observatory [online database]. Geneva: World Health Organization; (https://www.who.int/data/gho/data/themes/neglectedtropical-diseases, accessed on 26 March 2025).
- 25. Ending the neglect to attain the Sustainable Development Goals: a sustainability framework for action against neglected tropical diseases 2021–2030. Geneva: World Health Organization; 2021 (https://iris.who.int/ handle/10665/338886). Licence: CC BY-NC-SA 3.0 IGO.
- 26. WHO bacterial priority pathogens list, 2024: bacterial pathogens of public health importance to guide research, development and strategies to prevent and control antimicrobial resistance. Geneva: World Health Organization; 2024. Licence: CC BY-NC-SA 3.0 IGO.
- 27. GLASS dashboard [online database]. Geneva: World Health Organization; [in press]; (https://worldhealthorg.shinyapps. io/glass-dashboard/_w_ad3df80a/#!/home).
- United Nations Children's Fund, World Health Organization, International Bank for Reconstruction and Development/ The World Bank. Levels and trends in child malnutrition: key findings of the 2025 edition of the joint child malnutrition estimates. Geneva: World Health Organization; 2025 [in press]. Licence: CC BY-NC-SA 3.0 SA.
- 29. Discussion paper: 2025–2030 World Health Assembly global maternal, infant and young child nutrition targets and proposal for process indicators. Geneva: World Health Organization, 2024 (https://cdn.who.int/media/docs/defaultsource/nutrition-and-food-safety/discussion-paper-2025-2030-wha-nutrition-targets.pdf?sfvrsn=9fe91c03_8).
- WHO global anaemia estimates, 2025 edition. Geneva: World Health Organization; 2025 [in press]. (https://www.who.int/ data/gho/data/themes/topics/anaemia_in_women_and_ children).

- 31. WHO global report on trends in prevalence of tobacco use 2000–2030. Geneva: World Health Organization; 2024 (https:// iris.who.int/handle/10665/375711). Licence: CC BY-NC-SA 3.0 IGO.
- 32. Follow-up to the high-level meetings of the United Nations General Assembly on health-related issues: prevention and control of noncommunicable diseases: report by the Director-General. Geneva: World Health Organization; 2019 (A72/19; https://iris.who.int/handle/10665/328648).
- 33. Global alcohol action plan 2022–2030. Geneva: World Health Organization; 2024. (https://iris.who.int/ handle/10665/376939). Licence: CC BY-NC-SA 3.0 IGO
- 34. WHO global information system on alcohol and health. The global health observatory [online database]. Geneva: World Health Organization; 2025 [in press]. (https://www.who. int/data/gho/data/themes/global-information-system-on-alcohol-and-health).
- 35. Progress on household drinking water, sanitation and hygiene 2000–2022: special focus on gender. New York: United Nations Children's Fund and World Health Organization; 2023 (https://washdata.org/reports/jmp-2023-wash-households).
- 36. Progress on the proportion of domestic and industrial wastewater flows safely treated – mid-term status of SDG indicator 6.3.1 and acceleration needs, with a special focus on climate change, wastewater reuse and health. United Nations Human Settlements Programme and World Health Organization; 2024 (https://iris.who.int/ handle/10665/379601). Licence: CC BY-NC-SA 3.0 IGO.
- SDG 6.3.1 proportion of safely treated domestic wastewater flows. In: The Global health Observatory [website]. Geneva: World Health Organization; 2025 [in press]. (https://www. who.int/data/gho/data/indicators/indicator-details/GHO/ WSH_DOMESTIC_WASTE_SAFELY_TREATED).
- CRS: Creditor Reporting System (flows) [cloud replica].
 In: OECD data explorer [website]. Paris: Organisation for Economic Co-operation and Development; 2025 (https:// data-explorer.oecd.org/vis?df[ds]=DisseminateFinalBoost& df[id]=DSD_CRS%40DF_CRS&df[ag]=OECD.DCD.FSD&dq=DA C..1000.100._T._T.D.Q._T..&lom=LASTNPERIODS&lo=5&to[TI ME_PERIOD]=false, accessed on 22 April 2025).
- Air pollution data portal. The global health observatory [online database]. Geneva: World Health Organization; 2025 [in press]. (https://www.who.int/data/gho/data/themes/airpollution).
- 40. Air quality standards, second edition, 12 February 2025. World Health Organization; 2025 (https://www.who.int/tools/airquality-standards).
- 41. World population prospects 2024: special aggregates, UNrelated groups. Online edition. New York: United Nations Department of Economic and Social Affairs, Population Division; 2024 (https://population.un.org/wpp/).
- 42. Violence against women prevalence estimates, 2018: global, regional and national prevalence estimates for intimate partner violence against women and global and regional prevalence estimates for non-partner sexual violence against women. Geneva: World Health Organization; 2021 (https://iris.who.int/handle/10665/341337). Licence: CC BY-NC-SA 3.0 IGO.
- The Sustainable Development Goals report 2023: special edition: towards a rescue plan for people and planet. New York: United Nations; 2023 (https://unstats.un.org/sdgs/ report/2023/).

- Tracking universal health coverage: 2023 global monitoring report. Geneva: World Health Organization, International Bank for Reconstruction and Development / The World Bank; 2023 (https://iris.who.int/handle/10665/374059).
- Zhou B, Carrillo-Larco RM, Danaei G, Riley LM, Paciorek CJ, Stevens GA et al. Worldwide trends in hypertension prevalence and progress in treatment and control from 1990 to 2019: a pooled analysis of 1201 populationrepresentative studies with 104 million participants. Lancet. 2021;398(10304):957–980 (https://doi.org/10.1016/S0140-6736(21)01330-1).
- 46. Zhou B, Rayner AW, Gregg EW, Sheffer KE, Carrillo-Larco RM, Bennett JE et al. Worldwide trends in diabetes prevalence and treatment from 1990 to 2022: a pooled analysis of 1108 population-representative studies with 141 million participants. Lancet. 2024;404(10467):2077–2093 (https://doi. org/10.1016/S0140-6736(24)02317-1).
- 47. Estimates and projections of family planning indicators 2024. New York: United Nations, Department of Economic and Social Affairs, Population Division; 2024 (https://www.un.org/ development/desa/pd/data/family-planning-indicators).
- Immunization dashboard: global [WHO/UNICEF estimates of national immunization coverage (WUENIC), 2024 revision] [website]. Geneva: World Health Organization, United Nations Children's Fund; 2024 (https://immunizationdata.who.int/, accessed on 27 March 2025).
- 49. Delivery care: UNICEF/WHO joint database on births attended by skilled health personnel. New York: United Nations Children's Fund and World Health Organization; 2024.
- Immunization agenda 2030: a global strategy to leave no one behind. Geneva: World Health Organization; 2020 (https:// www.who.int/teams/immunization-vaccines-and-biologicals/ strategies/ia2030).
- Global survey on progress on SDG health target 3.5 (2019). [No publisher]; (https://cdn.who.int/media/docs/default-source/alcohol/monitoring/survey-health3-5-2019.pdf).
- 52. Global status report on alcohol and health and treatment of substance use disorders. Geneva: World Health Organization; 2024 (https://iris.who.int/bitstream/hand le/10665/377960/9789240096745-eng.pdf).
- 53. Data. In: The DHS Program [website]. (https://dhsprogram. com/data/, accessed on 27 March 2025).
- World health survey plus. World Health Organization; 2025 (https://www.who.int/data/data-collection-tools/worldhealth-survey-plus).
- 156th Executive Board, provisional agenda item 12, 20 December 2024. Health and care workforce: global strategy on human resources for health: workforce 2030. Report by the Director-General. Geneva: World Health Organization; 2024 (EB156/15; https://apps.who.int/gb/ebwha/pdf_files/EB156/ B156_15-en.pdf).
- National health workforce accounts data portal [online database]. Geneva: World Health Organization; (https://apps. who.int/nhwaportal).
- International Health Regulations (2005): States Parties selfassessment annual reporting tool, second edition. Geneva: World Health Organization; 2021 (https://cdn.who.int/media/ docs/default-source/health-security-preparedness/cap/ spar/9789240040120-eng-new.pdf). Licence: CC BY-NC-SA 3.0 IGO.
- Electronic IHR State Parties self-assessment annual reporting tool [online database]. Geneva: World Health Organization; (https://extranet.who.int/e-spar/, accessed on 27 March 2025).

- Global health expenditure database [online database].
 Geneva: World Health Organization; (https://apps.who.int/ nha/database/, accessed on 7 March 2025).
- 60. Official development assistance (ODA) for medical research and basic health sectors per capita, by recipient country. Global observatory on health research and development [website]. World Health Organization; (https://www.who.int/ observatories/global-observatory-on-health-research-anddevelopment/indicators/official-development-assistance-formedical-research-and-basic-health-sectors-per-capita-byrecipient-country, accessed on 7 March 2025).
- 61. SDG indicators: global indicator framework for the Sustainable Development Goals and targets of the 2030 Agenda for Sustainable Development. New York: United Nations, Department of Economic and Social Affairs; 2025 (https://unstats.un.org/sdgs/indicators/indicators-list/).
- 62. IAEG-SDGs 2025 comprehensive review process. New York: United Nations, Department of Economic and Social Affairs; 2025 (https://unstats.un.org/sdgs/iaeg-sdgs/2025comprehensive-review).

- Fifty-sixth session United Nations Statistical Commission, Provisional agenda item 3 (e). Inter-Agency and Expert Group on Sustainable Development Goal Indicators. Note by Secretary-General. New York: United Nations Economic and Social Council; 2024 (E/CN.3/2025/6; https://unstats.un.org/ UNSDWebsite/statcom/session_56/documents/2025-6-SDG-IAEG-E.pdf).
- Hogan DR, Stevens GA, Hosseinpoor AR, Boerma T. Monitoring universal health coverage within the Sustainable Development Goals: development and baseline data for an index of essential health services. Lancet Glob Health. 2017;6(2):E152–E168 (http://dx.doi.org/10.1016/S2214-109X(17)30472-2).
- Tracking universal health coverage: 2025 global monitoring report. Geneva: World Health Organization and International Bank for Reconstruction and Development / The World Bank; 2025 [in press]. Licence: CC BY-NC-SA 3.0 IGO.

Progress in achieving the Triple Billion targets

3

The Triple Billion targets, a cornerstone of WHO's Thirteenth General Programme of Work (GPW13), track the collective progress by WHO and its Member States in promoting, providing and protecting health worldwide. The latest update illustrates the progress made since the start of GPW13 in 2018 and the key challenges particularly in the universal health coverage and health emergencies protection billions. Anchored in the health-related SDGs, WHO's GPW13 provides a strategic roadmap to enhance health and wellbeing for all (1). More importantly, GPW13 championed the results framework with impact measurement at its core to reinforce organizational accountability and transparency. The foundation of the impact measurement is the 46 outcome indicators centred on promoting the overall health and quality of life for the people, providing essential health services, and protecting people from health emergencies. The Triple Billion targets give strategic clarity to the impact measurement of GPW13. By summarizing the outcome indicators into three ambitious, but easy to understand and communicate goals, the Triple Billion targets effectively convey global health priorities, motivate collective action by the global community, facilitate transparency and accountability, and complement the strategic priorities set by WHO's GPW13 to address current and emerging global health priorities.

For GPW13, the WHO Member States aim to enable one billion more people achieve better health and overall well-being, to provide one billion more people with access to essential health services without incurring financial hardship, and to better protect one billion more people from health emergencies worldwide by the end of the GPW13, extended from 2023 to 2025. This chapter provides a summary of the progress made in achieving the Triple Billion targets with projections to 2025, incorporating the latest available data from the outcome indicators at the country level.

3.1 Triple Billion targets progress

Progress towards achieving the Triple Billion targets remains uneven (Fig. 3.1). Growth in providing essential health services is insufficient, with only 431 million more people gaining affordable health-care access by 2024 relative to the GPW13 baseline year of 2018. This is projected to rise to 500 million by 2025 – half the targeted one billion. Furthermore, close to 637 million more individuals will be better protected from health emergencies by 2024. Despite significant progress, this is expected only to increase to 697 million people by 2025, around 30% short of the one-billion target. Remarkably, an estimated 1.4 billion more people will experience healthier lives by 2024, with 1.5 billion expected in 2025, surpassing the original target of one billion. Yet, such progress is not enough to put the world on track to achieve the health-related SDGs by 2030 (2,3). Moreover, recent interruptions in international aid threaten to disrupt health services and systems, disproportionally impacting countries and communities with the greatest health-care needs. Safeguarding the gains made in achieving the Triple Billion targets is a paramount task for the global community in the years ahead.



Figure 3.1 Progress in achieving the Triple Billion targets and estimated contributions by indicator

3.2 Progress in healthier populations

Current trends indicate that the world is on track to meet the goal of improving health and well-being for one more billion people by 2025. Compared with the baseline year of 2018, some 1.5 billion people are projected to be living healthier lives by 2025. Such progress is primarily due to reduced tobacco use, improved air quality, increased use of clean household fuels, and better access to water, sanitation and hygiene (Fig. 3.1). Yet, at the global level, significant challenges persist in the areas of rising rates of adult and childhood obesity.

Looking at the top contributing indicators by region (Fig. 3.2), the most striking features are worsening rates of adult obesity in the Easter Mediterranean Region and Region of the Americas. These account for setbacks equal to 32.5% and 62.1% of the respective regional net contributions. Despite these difficulties, substantial progress has been made. Most regions are seeing improvements in clean household fuel use as their primary marker of progress, with particularly large effects in the African Region (67.6%) and the South-East Asia Region (53.1%). In both these regions, improvements in clean fuel use represent over half of the additional people living healthier lives.

All regions showed substantial progress in safely managed sanitation, which ranked as the second most important contributor globally (32.7%). The top three indicators contributing to the Triple Billion targets in the European Region are improvements in air quality (38.4%), safely managed sanitation (26.1%) and tobacco use (22.8%). The African Region is unique in having safely managed water as a top contributor (14.9%), highlighting region-specific priorities and challenges in water infrastructure.



Figure 3.2 Top indicators with most contributions to the Triple Billion targets, 2018–2025^a

^a Percentage is calculated as the ratio of individual indicator contribution over magnitude of the overall contribution.

3.3 Progress in universal health coverage

Compared with the baseline year of 2018, about 500 million more people worldwide will enjoy access to essential health services without undue financial burden by 2025. This progress in achieving UHC has been supported by improvements in health workforce, enhanced coverage of hypertension treatment, and better access to family planning. However, the most substantial contributor to achieving UHC, and the progress made so far during the GPW13 period, is the expanded availability of antiretroviral therapy (ART) for HIV and AIDS. Challenges in access to health services continue to hinder faster progress, particularly in diabetes management and mitigating financial hardships incurred by utilizing health services. The declining coverage of DTP3 immunization is especially concerning.

Nearly all regions have seen progress in expanding access to HIV treatment (Fig. 3.2). However, there have been setbacks in progress in health security and affordability. In the Region of the Americas, the strong progress is substantially offset by deterioration in health security (-42.2%). Both the European and Western Pacific regions have seen significant increases in financial hardship. Rising financial hardship in the Western Pacific Region has significantly offset other UHC improvements, reducing progress by nearly 30%.

Most regions show contributions from multiple factors, indicating diverse approaches to managing health-care access. We see smaller, yet still substantial, gains in the health workforce in the Western Pacific Region (23.1%), as well as better tuberculosis treatment in the South-East Asia (17%) and Eastern Mediterranean regions (7.5%).

3.4 Progress in health emergencies protection

According to updated information for the outcome indicators, substantial advancements have been made towards better safeguarding an additional one billion people from health emergencies by 2025. Crucial amendments to the International Health Regulations, adopted by the World Health Assembly in 2024, represent a major step forward in global health security and pandemic preparedness. In addition, significant strides in early disease detection, mobilizing emergency response teams and ensuring equitable access to essential medical supplies, are key drivers of global resilience to health emergencies. Health emergencies protection shows the greatest amount of regional variation in top indicator contributions (Fig. 3.2). The Eastern Mediterranean Region reported exceptional progress in detection and response times (151.4%) but was undermined by worsening health security (-83.8%). Meanwhile, the Region of the Americas faces challenges in both health security (-94.9%) and prevention (-60.3%). The European Region made strong gains in detection and response, with a positive profile across indices (92% time to detect and respond, 7.8% health security). These contrasting outcomes highlight regional differences in emergency preparedness and response strategies.

3.5 Forecast to 2030 and scenarios

The Triple Billion target projections through 2030 showed encouraging trends towards a healthier global population. Under the current trajectory, 2.0 billion more people were expected to live healthier by 2030 compared with the 2018 baseline. Figure 3.3 presents an alternative scenario in which all countries meet the annualized targets set by the 2030 Agenda using the same set of 46 outcome indicators (4). In this case, 5.6 billion more people would live healthier lives by 2030 - nearly triple the gain of 2.0 billion under the current trajectory. Similarly, achieving the 2030 SDG and other global targets would result in 2.1 billion more people gaining access to UHC without financial hardship, compared with 0.8 billion under current trends. The gap is smaller for health emergency protection; meeting the 2030 targets would still safeguard 1.0 billion more people against health emergencies, compared with 0.8 billion under the current trajectory.

Figure 3.4 illustrates the potential for additional progress if within-region inequalities were to be addressed. It

shows the gap between forecasted contributions and a scenario where all countries are able to match their regional top performers by 2030. The large gaps highlight significant missed opportunities in progress both now and in the future.

The regional disparities are most striking for health emergencies protection. By 2030, an additional 30.6% of people in the Region of the Americas could be better protected if all countries matched their regional leaders – a sharp increase from 10.6% in 2025. Similarly, the African Region could extend protection to an additional 28% of its population, up from 9.6% in 2025.

For UHC, closing these gaps would mean an additional 13.8% of people in the European Region and over 15.5% more people in the Region of the Americas would have affordable access to essential health care. These figures underscore how reducing within-region disparities could dramatically accelerate progress toward the Triple Billion targets.



Figure 3.3 Progress needed to achieve the Triple Billion and 2030 global targets

Figure 3.4 Progress required for all countries to match the top performers in their region, on top of the Triple Billion target projections,^{a,b} 2025–2030



^a Matching the top performers means achieving the values of the indicators from the 10% best-performing countries in the same WHO region, by 2030. ^b Percentages are calculated as the ratio of additional persons required to match the top performers over the total 2030 population in the region.

Box 3.1 Transition from GPW13 to GPW14

The WHO GPW13 (2019–2023, extended to 2025), strengthened WHO's effectiveness and accountability in improving global health outcomes. Triple Billion targets, a key component of GPW13, were designed to enhance accountability and drive measurable health improvements worldwide. By strategically focusing on three target areas and leveraging existing data sources, WHO aimed to deliver tangible health benefits to billions of people.

In September 2023, the United Nations Heads of State and Government Meeting called for doubling the rate of progress on the SDGs to meet the 2030 Agenda *(5)*. In response, WHO is developing the fourteenth general programme of work (GPW14, 2025–2028) to address two fundamental challenges: slow progress and insufficient data.

Building on GPW13, GPW14 focuses on promoting, providing and protecting health while strengthening the global health ecosystem to advance the SDGs. Additionally, it seeks to enhance WHO's organizational performance across all three levels – global, regional and country.

The GPW14 results framework translates health goals into measurable targets through a structured results chain. This chain links WHO's work to broader health and development outcomes. As in GPW13, progress will be assessed using the Triple Billion indices through outcome indicators in three overarching areas: promote, provide and protect.

Triple Billion targets recalibration

Under GPW13, the Triple Billion targets for 2025 were defined in terms of relative changes – the additional number of people who achieved specific health status compared to 2018:

- 1 billion more people living healthier lives (promote)
- 1 billion more people benefitting from universal health coverage (provide)
- 1 billion more people better protected from health emergencies (protect)

For GPW14, the targets are recalibrated to reflect absolute population coverage by 2028, accounting for improved measurement methods and setting ambitious yet realistic goals:

- 6 billion people living healthier lives (promote)
- 5 billion people benefitting from universal health coverage without financial hardship (provide)
- 7 billion people protected from health emergencies (protect)

The revised Triple Billion targets offer several benefits:

- Better alignment with global targets, including the SDGs and World Health Assembly (WHA) resolutions
- Enable scenario analysis for assessing health interventions at regional and country levels
- Facilitate direct statistical forecasting using past trends and country-specific data
- Support benchmarking assessments to identify best practices at regional and global levels
- Balance ambition with realism, ensuring achievable yet aspirational goals
- Account for uncertainty in observed data and future projections
- Promote equity, ensuring that no one is left behind or unprotected.

3.6 Conclusion

The WHO Triple Billion framework has served as both a strategic guide and a practical tool for accountability and accelerating progress in global health. By providing a structured approach to tracking advancements in indicators across regions and countries, it offers decisionmakers actionable insights on where to focus resources and policy interventions. The 2025 projections, developed using the most recent available data, reveal both encouraging progress and gaps that require targeted action. Regional analysis shows how varied approaches to reaching these targets reflect different priorities, challenges and opportunities across the six WHO regions. Looking ahead, as WHO transitions from GPW13 to GPW14 in 2025, the core structure of the Triple Billion targets remains in place. However, the list of outcome indicators has been updated to reflect emerging health challenges and strategic priorities set collectively by the global community for GPW14 (see Box 3.1). Through GPW14, WHO reaffirms its unwavering commitment to addressing emerging global health priorities and tackling challenges, as mandated by its Member States, despite an evolving global landscape.

References

- Thirteenth general programme of work (GPW13): methods for impact measurement, version 2.1. Geneva: World Health Organization; 2020 (https://iris.who.int/handle/10665/341371). Licence: CC BY-NC-SA 3.0 IGO.
- 2. Thirteenth general programme of work (GPW13): metadata for impact measurement indicators. Geneva: World Health Organization; 2020 (https://www.who.int/publications/m/item/ metadata-for-impact-measurement-indicators).
- Seventieth session United Nations General Assembly, 25 September 2015. Transforming our world: the 2030 Agenda for Sustainable Development; 2015 (A/RES/70/1; https://docs. un.org/en/A/RES/70/1).
- 4. WHO results framework: delivering a measurable impact in countries. Geneva: World Health Organization; 2024 (https://www.who.int/publications/m/item/who-results-framework-delivering-a-measurable-impact-in-countries).
- Seventy-eighth session High-level Political Forum on Sustainable Development, 18 September 2023. New York, NY: United Nations; 2023 (GA/12529; https://press.un.org/en/2023/ ga12529.doc.htm).

Inequality in immunization

Ensuring equitable access to immunization is vital for achieving public health goals and ensuring that every person, regardless of their background and circumstances, fully benefits from vaccines to improve health and well-being. This chapter presents the state of inequality in immunization coverage. It shows how inequalities in immunization have changed in the last decade and discusses barriers to achieving equitable immunization coverage.

4.1 The importance of vaccine equity

Vaccines are life-saving interventions that have dramatically improved global health outcomes. Immunization currently prevents 3.5 million to 5 million deaths every year from diseases such as diphtheria, tetanus, pertussis, measles and influenza (1). Since the launch of the WHO Expanded Programme on Immunization in the mid-1970s, childhood vaccination coverage has improved significantly, contributing to a substantial reduction in U5MR (2). This underscores the critical role of vaccines in safeguarding public health and highlights the importance of continuing efforts to ensure equitable access to immunization worldwide.

Equity in immunization is a cornerstone of the Immunization Agenda 2030 (IA2030), which envisions a world where everyone, everywhere, at every age fully benefits from vaccines to improve health and well-being (3). This global strategy aims to leave no one behind by increasing equitable access to vaccines, ensuring that immunization programmes are integrated into primary health care and contributing to achieving universal health coverage. Global inequalities persist in vaccine access, financing, availability and distribution – both between countries and within countries. Low-income countries often face challenges in securing sufficient vaccine supplies due to limited financial resources, disruptions in health-care services, logistical challenges and inequities in access to services (4).

Within countries, inequalities in vaccine access and coverage can be influenced by factors such as geographic location, socioeconomic status and education levels. Rural and remote areas may have lower access to vaccination services compared with urban centres, and those affected by social disadvantage may face barriers to immunization due to lack of information or logistical challenges. Addressing inequalities requires targeted efforts to strengthen health systems, improve vaccine delivery infrastructure and ensure that immunization programmes are inclusive and accessible to all segments of the population.

4.2 Inequalities in immunization coverage persist within countries

The delivery of three doses of the combined DTP3 vaccine among children under the age of 1 year is used as a proxy indicator for monitoring the capacity of a health system to reliably reach children through routine vaccination channels. Measuring inequalities in DTP3 immunization coverage within countries reveals where gaps exist in a routinely delivered vaccine and helps to inform appropriate approaches to reach under-vaccinated populations. Figure 4.1 shows median inequality in DTP3 immunization coverage across LICs and middle-income countries with a recent household survey (from 2014–2023) by economic status, education and place of residence. For economic status and education, inequality is measured as the difference along the socioeconomic spectrum, while taking the composition of every socioeconomic subgroup into consideration.¹ For place of residence, inequality is measured as the difference as the difference between urban and rural subgroups. Medians are calculated across all countries

with data available, as well as by World Bank country income grouping.

The results indicate inequalities favouring those living in richer households (median economic-related inequality of 10.4 percentage points across 88 countries), having more education (median education-related inequality of 18.7 percentage points across 67 countries) and living in urban areas (median difference of 2.3 percentage points across 89 countries). Economic-related inequality in DTP3 immunization coverage is most pronounced in LICs and LMICs, where there are median inequalities of 13.7 and 13.0 percentage points in coverage along the economic status spectrum, respectively. Educationrelated inequality was high across UMICs, LMICs and LICs, while differences between urban and rural areas were low overall across country income groupings.

Figure 4.1 Median economic-related, education-related and place of residence inequality in DTP3 immunization coverage, by World Bank country income group, 2014–2023



DTP3: diphtheria-tetanus-pertussis vaccine third dose. The median difference for economic status and education reflects the difference along the whole socioeconomic spectrum, while taking the composition of every socioeconomic subgroup into consideration. The median difference for place of residence reflects the difference between urban and rural subgroups.

Source: derived from the WHO Health Inequality Data Repository, Childhood Immunization dataset, with data sourced from the most recent Demographic and Health Survey or Multiple Indicator Cluster Survey between 2014 and 2023 (5).

The number of children who have not received any routine vaccines (referred to as zero-dose children) reflects the accessibility of basic health services, including immunization services. The IA2030 includes an objective to extend immunization services to regularly reach zero-dose children, with a target of a 50% reduction in zero-dose children between 2019 and 2030 (*3*). An analysis of inequalities in the proportion of 1-yearold children who had not received any doses of the DTP vaccine, using household survey data across 88 countries, reveals substantial economic- and education-related inequalities (Fig. 4.2). Data from 87 countries showed a median economic-related inequality of 6.7 percentage points in zero-dose DTP prevalence, but inequalities were particularly pronounced across LMICs and LICs, where median economic-related inequality was greater than 9 percentage points. Across 66 countries, median education-related inequality in zero-dose prevalence was 10.5 percentage points but inequalities were 16 percentage points in LMICs and LICs. Overall, there was low median inequality between rural and urban areas in 88 countries, but zero-dose prevalence was slightly higher in rural than urban areas in LMICs and LICs.

Figure 4.2 Median economic-related, education-related and place of residence inequality in zero-dose DTP prevalence, by World Bank country income group, 2014–2023



DTP3: diphtheria-tetanus-pertussis vaccine third dose. The median difference for economic status and education reflects the difference along the whole socioeconomic spectrum, while taking the composition of every socioeconomic subgroup into consideration. The median difference for place of residence reflects the difference between rural and urban subgroups.

Source: derived from the WHO Health Inequality Data Repository, Childhood Immunization dataset, with data sourced from the most recent Demographic and Health Survey or Multiple Indicator Cluster Survey between 2014 and 2023 (5).

Eliminating economic-related inequality in DTP3 coverage and zero-dose prevalence has the potential for many more children to benefit from immunization. The potential for improvement in each country is premised on all children achieving the same outcome as the richest 20% of households. If economic-related inequality in DTP3 coverage were eliminated across the 88 countries with household survey data from 2014–2023, the weighted national average across these countries would improve from 76.0% to 85.5% (Fig. 4.3). If countries reduced the prevalence of zero-dose children to that of the richest subgroup, the weighted national average prevalence would be more than halved, from 13.1% to 6.3%.

Figure 4.3 Potential improvement in national average DTP3 coverage and zero-dose children by eliminating economic-related inequality, 88 countries, 2014–2023



DTP3: three doses of diphtheria-tetanus-pertussis vaccine.

The potential for improvement (dark blue vertical line) represents the overall weighted average that would be possible if, in each country, all 1-year-olds had the same level of coverage as the most advantaged subgroup (richest quintile). The current weighted average is indicated by the light blue bar. *Source*: derived from the WHO Health Inequality Data Repository, Childhood Immunization dataset, with data sourced from the most recent Demographic and Health Survey or Multiple Indicator Cluster Survey between 2014 and 2023 (5).
Of 88 countries with household survey data from 2014–2023, only 27 had DTP3 coverage of 90% or more, while coverage was 80–89% in 20 countries, 70–79% in 12 countries and below 70% in 29 countries (Fig. 4.4A). After considering the potential improvement in national average by eliminating economic-related inequality, 6 additional countries would have coverage greater than 90% and 10 additional countries would have coverage of 80–89%. If economic-related inequality in zero-dose prevalence were eliminated, 16 additional countries would reduce zero-dose prevalence to below 5% (Fig. 4.4B). The link between economic status and immunization coverage is complex and, as Box 4.1 shows, inequalities can emerge at various points of contact with the health system – underscoring the need for exploration of the drivers of inequalities and context-specific strategies to address them.

Figure 4.4 Potential improvement in national average by eliminating economic-related inequality, 88 countries, 2014–2023



The potential improvement reflects all 1-year-olds having the same outcome as the richest 20% of the population.

Source: derived from the WHO Health Inequality Data Repository, Childhood Immunization dataset, with data sourced from the most recent Demographic and Health Survey or Multiple Indicator Cluster Survey between 2014 and 2023 (5).

Box 4.1 Inequalities can be introduced through interventions: the example of tetanus immunization coverage at birth

Maternal and neonatal tetanus (MNT) is a form of tetanus that affects women during pregnancy or within six weeks of the end of pregnancy, and infants during their first 28 days of life (6). MNT constitutes a major public health concern, as neonatal case-fatality rates are upwards of 80% and approach 100% when untreated (7). Immunization of pregnant women against tetanus is a key strategy for reducing tetanus morbidity and mortality while also achieving the goal of maternal and neonatal tetanus elimination. An infant's tetanus protection at birth (PAB) is comprised of immunization received by the woman during and before her pregnancy (e.g. through childhood vaccination, booster doses or mass vaccination campaigns, or during prior pregnancies).

Global inequalities in tetanus immunization coverage at birth are evident across various dimensions, including wealth, maternal age, education and place of residence (8). For instance, poorer households, younger maternal age, lower education of the mother and residence in rural areas tend to have lower rates of PAB. These inequalities are particularly pronounced in LICs and LMICs, but they also persist in UMICs, especially concerning education level of the mother and rural residence.

Findings from a study of 71 LICs and middle-income countries suggest that the majority of tetanus immunization PAB for first births is the result of vaccinations received during pregnancy. The extent of economic-related inequality in PAB increases during pregnancy, as more women from richer households than women from poorer households are vaccinated during pregnancy (9). Economic-related inequality is measured as the difference along the socioeconomic spectrum, while taking the composition of every socioeconomic subgroup into consideration. Overall, LICs and middle-income countries had similar levels of economic-related inequality and tetanus immunization coverage before pregnancy. However, increases in tetanus immunization coverage at birth were coupled with increases in inequality in LICs, overall, in contrast with UMICs (Fig. 4.5). Inequalities in coverage at birth in this group were introduced during pregnancy, likely driven in part by inequalities in antenatal care access and utilization. Addressing these inequalities requires targeted interventions to improve access to and uptake of tetanus vaccines among women from poorer households, ensuring that all pregnant women and newborns receive adequate protection against this preventable disease.

Figure 4.5 Economic-related inequality in maternal tetanus immunization coverage and average coverage level before pregnancy and at birth, by World Bank country income group, 71 countries, 2013–2022



The start of the arrow indicates the level of economic-related inequality and tetanus immunization coverage before pregnancy and the end of the arrow indicates the level of inequality and coverage at birth. For each country, economic-related inequality is measured as the difference along the whole socioeconomic spectrum while taking the composition of every socioeconomic subgroup into consideration. For each World Bank income group, the weighted mean difference is calculated across countries. Based on 18 upper-middle-income, 33 lower-middle-income and 20 low-income countries. *Source*: derived from Johns et al. (9) with data sourced from the most recent Demographic and Health Survey or Multiple Indicator Cluster Survey between 2013 and 2022.

4.3 How inequalities have changed over time

Over the past two decades, significant efforts have been made to address inequalities in immunization coverage. IA2030 and its predecessor, the Global Vaccine Action Plan for 2011–2020, focus on reaching zero-dose children and ensuring equitable access to vaccines. Collaborations between WHO, UNICEF and Gavi, the Vaccine Alliance, have significantly impacted the number of zero-dose children in lower-income countries. For example, these efforts have helped to reduce the number of zero-dose children from 15.8 million in 2010 to 12.8 million in 2019 (10). However, the COVID-19 pandemic resulted in setbacks to vaccination programmes, resulting in an increase in the number of zero-dose children to 14.5 million in 2023. In response to disruptions caused by the COVID-19 pandemic, WHO and UNICEF launched catch-up and recovery programmes to restore immunization services. Targeted vaccination campaigns

in low-coverage regions aim to close immunization gaps and prevent outbreaks.

The overall change in inequality in DTP3 immunization coverage and zero-dose prevalence can be measured using the median difference along the socioeconomic spectrum (while taking the composition of every socioeconomic subgroup into consideration). Between 2004–2013 and 2014–2023, overall economic- and education-related inequalities in DTP3 immunization coverage decreased (Fig. 4.6). Across 72 countries, economic-related inequality in DTP3 coverage reduced from a median of 15.4 to 11.0 percentage points, and education-related inequality across 56 countries reduced from a median of 22.1 to 18.7 percentage points. Overall, economic- and education-related inequalities in zerodose prevalence only decreased slightly over this period.

Figure 4.6 Change in median economic, education and place of residence inequality in DTP3 immunization coverage and zero-dose prevalence between 2004–2013 and 2014–2023, by World Bank country income group



DTP3: three doses of diphtheria-tetanus-pertussis vaccine.

The start of the arrow indicates the median level of inequality in 2004–2013 and the end of the arrow indicates the median level of inequality in 2014–2023. The median difference for economic status and education reflects the difference along the whole socioeconomic spectrum, while taking the composition of every socioeconomic subgroup into consideration. The median difference for place of residence reflects the difference between urban and rural subgroups for DTP3 coverage and between rural and urban subgroups for zero-dose prevalence. Blue indicates a decrease in inequality, while orange indicates an increase in inequality. *Source*: derived from the WHO Health Inequality Data Repository, Childhood Immunization dataset, with data sourced from the most recent Demographic and Health Survey or Multiple Indicator Cluster Survey between 2014 and 2023 and between 2004 and 2013 (5).

Changes in immunization inequality related to economic status, education and place of residence varied across country income groupings. Broadly, in the last decade there has been a convergence in inequalities across LICs and LMICs, with inequalities diminishing in LICs to the point that their median level of inequality in 2014–2023 was similar to (or better than) that of LMICs. In particular, across 19 LICs, economic-related inequality in DTP3 coverage and zero-dose prevalence reduced by more than half, from 26.2 to 11.8 percentage points and from 17.2 to 8.0 percentage points, respectively. In UMICs, inequality remained the same for zero-dose prevalence but economic- and education-related inequality in DTP3 coverage increased.

4.4 Barriers to immunization

The Equity Reference Group for Immunization² has identified several thematic priorities to address inequities in immunization coverage, including urban poor areas, remote/rural areas, children affected by conflict, and gender-related inequities and barriers (11). In such contexts, different strategies are needed to understand coverage levels and address inequities in immunization coverage. Surveys are rarely carried out to enable the accurate assessment of coverage rates in the above contexts, and when they are, they generally lack granularity. This results in a scarcity of disaggregated data and inadequate monitoring of inequalities. Consequently, it is challenging to identify disadvantaged groups within these contexts and implement programmes that prioritize reaching them.

In urban poor areas, overcrowding and poverty can significantly hinder vaccine delivery and uptake. These areas often lack adequate health-care facilities and resources, making it challenging to maintain consistent immunization services. Remote rural regions face similar issues, with limited health-care infrastructure and difficult terrain further complicating efforts to reach all children. For instance, a study in Malawi found that proximity to a health facility providing vaccination services in rural areas was associated with increased likelihood of vaccine receipt, and that remote rural children were more likely to be under-vaccinated (12). Additionally, children affected by conflict experience unique barriers to immunization. Conflict zones often see disrupted health services, displacement of populations and heightened insecurity, all of which contribute to lower immunization rates. These children may miss out on routine vaccinations due to instability and lack of access to health care.

Gender-related barriers can hinder immunization coverage by limiting women's autonomy and decisionmaking power, which affects their ability to access health services for themselves and their children. Additionally, cultural norms and practices may restrict women's mobility and access to information, further reducing vaccination coverage in communities with high gender inequality. Women with higher social independence, as measured by the Survey-based Women's Empowerment Global Index, are more likely to ensure their children receive vaccinations (13). In contrast, DTP3 coverage tends to be lower, and zero-dose prevalence higher, among children of women with lower social independence.

At the subnational and national levels, areas with higher gender inequality tend to have lower immunization coverage. In a study of 162 countries, those with high gender inequality (favouring men), as measured using the Gender Inequality Index,³ had DTP3 immunization coverage that was 13 percentage points lower (median value of 81% versus 94%) and zero-dose prevalence 7 percentage points higher (median value of 10% versus 3%) than countries with lower inequality (14). A study of 702 subnational regions across 57 countries also found that regions with greater gender inequality had lower DTP3 coverage and higher zero-dose prevalence (15). Subnational regions with higher gender inequality (favouring men), as measured by the Subnational Gender Development Index, had DTP3 immunization coverage that was 21.6 percentage points lower, while zero-dose prevalence was 13.4 percentage points higher (median values).

² The Equity Reference Group for Immunization is an action-oriented think tank that consists of senior experts in global health working with WHO, Gavi, the World Bank, Bill & Melinda Gates Foundation and UNICEF; academic experts in metrics, gender, and health systems development; and senior leaders from the ministries of health in Indonesia, Ethiopia and the United Republic of Tanzania.

³ The Gender Inequality Index is a composite metric of gender inequality using three dimensions: reproductive health, empowerment and the labour market. It shows the loss in potential human development due to inequality between female and male achievements in these dimensions.

4.5 Data availability to monitor inequalities in immunization

Monitoring inequalities in immunization relies on various data sources, each with its own advantages and disadvantages (16). Administrative data are routinely collected by health facilities and can provide timely, comprehensive information on immunization coverage. These data are cost-effective and cover large populations, making them useful for tracking trends over time. However, administrative data can suffer from inaccuracies due to reporting errors, incomplete records, sparse information about dimensions of inequality and uncertainty about target population numbers. Household survey data, such as those from DHS and Multiple Indicator Cluster Surveys (MICS), offer detailed, population-based insights into immunization coverage. These surveys are valuable for capturing data on hardto-reach populations and for validating administrative data. They also provide rich contextual information, such as socioeconomic and demographic factors. However, household surveys are expensive, time-consuming and conducted infrequently, which can limit their utility for real-time monitoring.

Data availability can be a challenge, particularly in LICs where resources for comprehensive data collection and reporting are often limited. These countries may face difficulties in maintaining accurate and up-todate records, which can hinder efforts to monitor and address immunization inequalities effectively. Logistical challenges, such as limited access to remote areas and insufficient funding for large-scale surveys, further complicate data collection efforts. Inaccurate estimates of the target population for vaccination can exacerbate these difficulties, particularly when monitoring subnational inequalities and subpopulations of interest. Misestimations can lead to skewed coverage rates, masking inequalities and making it challenging to identify and address areas with the greatest need (*17*). Addressing these challenges requires investment in strengthening health information systems to ensure reliable data are available for informed decision-making and targeted interventions.

A scoping review of publications in academic journals analysing inequalities in immunization coverage among children revealed a rise in the use of DHS and MICS data between 2014 and 2023 (Fig. 4.7) (18). Across the 10-year period, 129 papers (54%) had utilized DHS and/or MICS data, demonstrating the utility of these data sources for monitoring inequalities in immunization. The number of publications using administrative, health or surveillance records remained relatively stable, as did those based on other sources such as non-routine, study-specific or small-scale surveys.



Figure 4.7 Publications of studies of inequalities in childhood vaccination coverage between 2014 and 2023 by data source for vaccination indicators

Other sources

References

- Vaccines and immunization [website]. World Health Organization; 2025 (https://www.who.int/health-topics/ vaccines-and-immunization/, accessed on 17 February 2025).
- Shattock AJ, Johnson HC, Sim SY, Carter A, Lambach P, Hutubessy RCW et al. Contribution of vaccination to improved survival and health: modelling 50 years of the Expanded Programme on Immunization. Lancet. 2024;403(10441):2307– 2316 (https://doi.org/10.1016/S0140-6736(24)00850-X).
- Immunization agenda 2030: a global strategy to leave no one behind. Geneva: World Health Organization; 2020 (https:// www.who.int/publications/m/item/immunization-agenda-2030-a-global-strategy-to-leave-no-one-behind, accessed 11 March 2025).
- Vaccine inequity undermining global economic recovery. Joint news release, 22 July. World Health Organization; 2021 (https://www.who.int/news/item/22-07-2021-vaccineinequity-undermining-global-economic-recovery).
- 5. Health Inequality Data Repository [database]. World Health Organization; 2025 (https://www.who.int/data/inequalitymonitor/data, accessed on 17 February 2025).
- Roper MH, Vandelaer JH, Gasse FL. Maternal and neonatal tetanus. Lancet. 2007;370(9603):1947–1959 (https://doi. org/10.1016/S0140-6736(07)61261-6).
- World Health Organization. Tetanus vaccines: WHO position paper – February 2017. Wkly Epidemiol Rec. 2017;6(92):53–76 (https://iris.who.int/handle/10665/254583).
- Johns NE, Cata-Preta BO, Kirkby K, Arroyave L, Bergen N, Danovaro-Holliday MC et al. Inequalities in immunization against maternal and neonatal tetanus: a cross-sectional analysis of protection at birth coverage using household health survey data from 76 countries. Vaccines. 2023;11(4):752 (https://doi.org/10.3390/vaccines11040752).
- Johns NE, Blumenberg C, Kirkby K, Allorant A, Costa FDS, Danovaro-Holliday MC et al. Comparison of wealth-related inequality in tetanus vaccination coverage before and during pregnancy: a cross-sectional analysis of 72 low- and middleincome countries. Vaccines. 2024;12(4):431 (https://doi. org/10.3390/vaccines12040431).
- Jones CD, Danovaro-Holliday C, Mwinnyaa G, Gacic-Dobo M, Francis L, Grevendonk J et al. Routine vaccination coverage – worldwide 2023. Wkly Epidemiol Rec. 2024;99:44 (https://iris. who.int/handle/10665/379414).
- 11. ERG products [website]. Equity Reference Group for Immunization; 2025 (https://sites.google.com/view/ erg4immunisation/products, accessed on 17 February 2025).

- Johns NE, Hosseinpoor AR, Chisema M, Danovaro-Holliday MC, Kirkby K, Schlotheuber A et al. Association between childhood immunisation coverage and proximity to health facilities in rural settings: a cross-sectional analysis of Service Provision Assessment 2013–2014 facility data and Demographic and Health Survey 2015-2016 individual data in Malawi. BMJ Open. 2022;12(7) (https://doi.org/10.1136/ bmjopen-2022-061346).
- Johns NE, Santos TM, Arroyave L, Cata-Preta BO, Heidari S, Kirkby K et al. Gender-related inequality in childhood immunization coverage: a cross-sectional analysis of DTP3 coverage and zero-dose DTP prevalence in 52 countries using the SWPER Global Index. Vaccines. 2022;10(7):988 (https://doi. org/10.3390/vaccines10070988).
- Fuertes CV, Johns NE, Goodman TS, Heidari S, Munro J, Hosseinpoor AR. The association between childhood immunization and gender inequality: a multi-country ecological analysis of zero-dose DTP prevalence and DTP3 immunization coverage. Vaccines. 2022;10(7):1032 (https:// doi.org/10.3390/vaccines10071032).
- Johns NE, Kirkby K, Goodman TS, Heidari S, Munro J, Shendale S et al. Subnational gender inequality and childhood immunization: an ecological analysis of the Subnational Gender Development Index and DTP coverage outcomes across 57 countries. Vaccines. 2022;10(11):1951 (https://doi.org/10.3390/vaccines10111951).
- Health inequality monitoring: harnessing data to advance health equity. Geneva: World Health Organization; 2024 (https://iris.who.int/handle/10665/379703). Licence: CC BY-NC-SA 3.0 IGO.
- Hosseinpoor AR, Bergen N, Barros AJ, Wong KL, Boerma T, Victora CG. Monitoring subnational regional inequalities in health: measurement approaches and challenges. Int J Equity Heal. 2016;15(1):1–13 (https://doi.org/10.1186/s12939-016-0307-y).
- Lyons C, Nambiar D, Johns NE, Allorant A, Bergen N, Hosseinpoor AR. Inequality in childhood immunization coverage: a scoping review of data sources, analyses, and reporting methods. Vaccines. 2024;12(8):850 (https://doi. org/10.3390/vaccines12080850).

Annexes

Annex 1.

Progress assessment of selected health-related SDG indicators by WHO region

A summary of progress of selected health-related SDG indicators is presented in Table A1.1 for Goal 3 (Health and Well-being) indicators and Table A1.2 for those outside of Goal 3.

An indicator is included in the assessment if it meets the following criteria:

- There is an explicit numerical SDG or other global framework target to be achieved by a defined year, applicable at both global and regional levels.
- Estimates are available for the baseline year and at least one other later year to enable a trend assessment.

For each indicator, the numbers shown in the global and regional cells are the latest available estimated values. The reference year of these values is indicated in the corresponding column.

Progress is assessed based on the latest available value, and the projected value for the target year based on

current trends. Unless otherwise noted, current trends are defined by the average annual rates of change between the baseline year and the reference year of the latest available value. Progress is described in three categories as follows:

- "Target reached": target is already achieved
- "Progress on track": target will be achieved by the target year based on current trends
- "Acceleration needed": target will not be met by the target year based on current trends

It is noted that this method of assessment may differ from those used by the United Nations Statistics Division or by the custodian agencies of individual indicators, leading to different assessment results.

Tables A1.3 and A1.4 provide the list of health-related SDG indicators that are not included in Tables A1.1 and A1.2.

Table A1.1 Latest values^a and progress assessment^b of selected SDG 3 indicators,^c by WHO region

			Year					seo	t t		ueəu	noig
Indicator	Notes	Baseline	Target	Most recent estimate	Target	lsdolð	nธวiาใA noig9Я	fo noig9Я the Ameri	263-dtuo2 Dig9Я si2A	European Region	Eastern Mediterra Region	Western Pacific Re
3.1.2 Proportion of births attended by skilled health personnel (%)		2015	2025	2024	06	87	74	96	06	66	85	86
3.3.1 New HIV infections (per 1000 uninfected population)	σ	2010	2030	2023	Reduce new infections by 90%	0.17	0.55	0.16	0.06	0.18	60.0	0.07
3.3.2 Tuberculosis incidence (per 100 000 population)		2015	2030	2023	Reduce by 80%	134	206	33	234	24	116	97
3.3.3 Malaria incidence (per 1000 population at risk)		2015	2030	2023	Reduce by 90%	60.4	226.8	3.6	2.3	0.0	17.9	2.3
3.3.4 Hepatitis B surface antigen (HBsAg) prevalence among children under 5 years (%)	Φ	2015	2030	2020	0.1	0.9	2.5	0.1	0.4	0.3	0.8	0.3
 3.3.5 Reported number of people requiring interventions against NTDs (millions) 	f	2010	2030	2023	Reduce by 90%	1495	518.6	30.7	7.87.7	5.9	78.0	74.0
3.4.1 Probability of dying between exact ages 30 and 70 from any of cardiovascular disease, cancer, diabetes or chronic respiratory illness (%)		2015	2030	2019	Reduce by 1/3	18.0	21.0	13.9	22.9	16.3	22.7	15.7
3.4.2 Suicide mortality rate (per 100 000 population)		2015	2030	2021	Reduce by 1/3	9.2	7.3	6.6	10.1	12.4	3.6	9.5
3.5.2 Total alcohol per capita (aged 15 years and older) consumption (litres of pure alcohol)		2010	2030	2022	Reduce by 20%	5.0	3.5	8.0	3.6	9.1	0.2	5.2
3.6.1 Death rate due to road traffic injuries (per 100 000 population)	0.0	2021	2030	2021	Halve the number of deaths	14.9	19.5	14.2	15.4	6.8	16.3	15.4
3.a.1 Age-standardized prevalence of current tobacco use among persons 15 years and older (%)	ч	2010	2025	2022	Reduce by 30%	20.9	9.5	16.6	26.5	25.3	17.9	22.5
3.b.1 Diphtheria-tetanus-pertussis third-dose (DTP3) immunization coverage among 1-year-olds (%)		2019	2030	2023	06	84	74	86	06	95	62	92
3.b.1 Measles-containing-vaccine second-dose (MCV2) immunization coverage by the locally recommended age (%)		2019	2030	2023	06	74	49	75	85	91	73	06
3.b.1 Pneumococcal conjugate vaccine third-dose (PCV3) immunization coverage among 1-year-olds (%)		2019	2030	2023	06	65	70	76	75	86	52	26
3.b.1 Human papillomavirus (HPV) immunization coverage estimates among 15-year-old girls (%)		2019	2030	2023	06	20	32	55	œ	35	2	7
NTDs: neglected tropical diseases. Notes:		f Project in the r	tion is based next years.	l on multi-diseas	e country-level data an	d planned int	erventions				Target n	eached
 The numbers shown in the global and regional cells are the latest available estimated values. Progress is assessed on the basis of the latest available value, and the projected value for the target year based on the average rates of change since the baselin 	р а	g Projec of deat Action h No reg	tion for the Carles the Source 201 for Road Sa jon or count	2030 number of (10, which was th fety 2010–2020. rry will be assess	leaths is based on the c e baseline of the first Un ed as "Target reached"	hange in the r iited Nations I before 2025 d	number Decade of ata are				Progres	s on track tion needed
year, unless otherwise stated.		availat	ole.									

Annexes

Progress assessment is based on the change in the total number of new infections. Projection takes into account preliminary updates to the estimates.

later year.

J

An SDG 3 indicator is included if: (1) there is an explicit numerical SDG or other global framework target to be achieved by a defined year, applicable at global and regional levels, and (2) data are available for the baseline year and at least one other

i Projection takes into account recent introduction of the vaccine in several countries. available.

Source: Births attended by skilled health personnel (1), HIV (2,3), tuberculosis (4), malaria (5), hepatitis (6), NTD (7), noncommunicable disease premature mortality (8), suicide mortality (8, 10), tobacco (11), immunization (12).

e

σ

Table A1.2 Latest values^a and progress assessment^b of selected health-related SDG indicators (excluding Goal 3),^c by WHO region

			Year					icas	ts t		ueəue	นอเฮล
Indicator	Notes	Baseline	Target	Most recent estimate	⁻ Target	Jedolð	nsiritan noig9Я	lo noigəЯ 19mA ədt	в∃-dtuo2 ig9Я si2A	Europear Region	Eastern Mediterra Region	Western Pacific Re
2.2.1 Prevalence of stunting among children under 5 years of age (%)	q	2012	2030	2024	Halve the number of stunted children	23.2	31.7	6.6	29.7	4.6	26.0	10.2
2.2.2 Prevalence of wasting among children under 5 years of age (%)	p	2012	2030	2024	e	6.6	5.6	0.7	13.9	1.0	6.3	1.9
2.2.2 Prevalence of overweight among children under 5 years of age (%)	σ	2012	2030	2024	m	5.5	4.2	9.1	3.3	7.5	4.5	9.7
2.2.3 Prevalence of anaemia in women aged 15 to 49 years (%)	σ	2012	2030	2023	Reduce by 50%	30.7	36.5	18.3	46.4	21.3	35.8	16.5
6.1.1 Proportion of population using safely managed drinking water services (%)		2015	2030	2022	66	73	33	81		92	67	
6.2.1 Proportion of population using safely managed sanitation services (%)		2015	2030	2022	66	57	26	66	49	78	55	68
6.2.1 Proportion of population using a handwashing facility with soap and water (%)	a	2015	2030	2022	66	75	26		76		72	95
7.1.2 Proportion of population with primary reliance on clean fuels and technology (%)		2015	2030	2023	66	74	24	93	73	97	74	87
Notes: a The numbers shown in the global and regional cells are the latest availab	ele	q	1ethodology owards the 2	based on the WI 030 global nutri	HO–UNICEF Meth tion targets. Glok	odology for m sal Nutrition Ta	onitoring prog Irgets Tracking	ress J Tool			Targe	st reached

- estimated values.
- Progress is assessed on the basis of the latest available value, and the projected value for the target year based on the average rates of change since the baseline year, unless otherwise stated. 9
 - An indicator is included if: (1) there is an explicit numerical SDG or other global framework target to be achieved by a defined year, applicable at global and regional levels, and (2) data are available for the baseline year and at least one other later year. υ
- Projection for the Western Pacific Region is based on the average annual rate https://www.who.int/tools/global-targets-tracking-tool. Ð
 - of change since 2017, the earliest year for the which the regional estimate is available.

Acceleration needed Insufficient data to

assess progress

Progress on track

Source: Child malnutrition (13), anaemia (14), water, sanitation and hygiene (15), clean fuels (16).

Table A1.3 List of SDG 3 indicators not included in the assessment

Indicator	Reason for exclusion
3.1.1 Maternal mortality ratio	Target set at the global level; progress not assessed at the regional level
3.2.1 Under-5 mortality rate	Target set at the country level; progress not assessed at the regional level
3.2.2 Neonatal mortality rate	Target set at the country level; progress not assessed at the regional level
3.5.1 Coverage of treatment interventions (pharmacological, psychosocial and rehabilitation and aftercare services) for substance use disorders	No explicit numerical target
3.7.1 Proportion of women of reproductive age (aged 15–49 years) who have their need for family planning satisfied with modern methods	No explicit numerical target
3.7.2 Adolescent birth rate	No explicit numerical target
3.8.1 Coverage of essential health services	No explicit numerical target
3.8.2 Proportion of population with large household expenditures on health as a share of total household expenditure or income	No explicit numerical target
3.9.1 Mortality rate attributed to household and ambient air pollution	No explicit numerical target
3.9.2 Mortality rate attributed to unsafe water, unsafe sanitation and lack of hygiene	No explicit numerical target
3.9.3 Mortality rate attributed to unintentional poisoning	No explicit numerical target
3.b.3 Health product access index	No explicit numerical target
3.c.1. Health worker density and distribution	No explicit numerical target
3.d.1 International Health Regulations (IHR) capacity and health emergency preparedness	No explicit numerical target
3.d.2 Percentage of bloodstream infections due to selected antimicrobial- resistant organisms	No explicit numerical target

Table A1.4 List of health-related SDG indicators (excluding Goal 3) not included in the assessment

Indicator	Reason for exclusion
1.a.2 Domestic general government health expenditure (GGHE-D) as percentage of general government expenditure (GGE) (%)	No explicit numerical target
5.2.1 Proportion of ever-partnered women and girls aged 15 years and older subjected to physical, sexual or psychological violence by a current or former intimate partner in the previous 12 months (%)	Trend data not available
5.2.2 Proportion of women and girls aged 15 years and older subjected to sexual violence by persons other than an intimate partner in lifetime (%)	Trend data not available
6.3.1 Proportion of domestic and industrial wastewater flows safely treated (%)	Baseline data not available
11.6.2 Annual mean concentrations of fine particulate matter (PM2.5) in urban areas $(\mu g/m^3)$	No explicit numerical target
16.1.1 Mortality rate due to homicide (per 100 000 population)	No explicit numerical target

References

- 1. Delivery care: UNICEF/WHO joint database on births attended by skilled health personnel. New York: United Nations Children's Fund and World Health Organization; 2024.
- 2. HIV. The global health observatory [online database]. Geneva: World Health Organization; (https://www.who.int/data/gho/ data/themes/hiv-aids, accessed on 26 March 2025).
- 3. AIDSinfo [online database]. Geneva: Joint United Nations Programme on HIV/AIDS (UNAIDS); (http://aidsinfo.unaids. org, accessed on 26 March 2025).
- Global tuberculosis report 2024. Geneva: World Health Organization; 2024 (https://iris.who.int/ handle/10665/379339). Licence: CC BY-NC-SA 3.0 IGO.
- World malaria report 2024: addressing inequity in the global malaria response. Geneva: World Health Organization; 2024 (https://iris.who.int/handle/10665/379751). Licence: CC BY-NC-SA 3.0 IGO.
- Global hepatitis report 2024: action for access in low- and middle-income countries. Geneva: World Health Organization; 2024 (https://iris.who.int/handle/10665/376461). Licence: CC BY-NC-SA 3.0 IGO.
- Neglected tropical diseases. The global health observatory [online database]. Geneva: World Health Organization; (https://www.who.int/data/gho/data/themes/neglectedtropical-diseases, accessed on 26 March 2025).
- 8. Global health estimates 2021. Geneva: World Health Organization; 2024 (https://www.who.int/data/global-healthestimates, accessed on 19 February 2025).
- 9. WHO global information system on alcohol and health. The global health observatory [online database]. Geneva: World Health Organization; 2025 [in press]. (https://www.who. int/data/gho/data/themes/global-information-system-on-alcohol-and-health).

- 10. Global status report on road safety 2023. Geneva: World Health Organization; 2023 (https://iris.who.int/ handle/10665/375016). Licence: CC BY-NC-SA 3.0 IGO.
- 11. WHO global report on trends in prevalence of tobacco use 2000–2030. Geneva: World Health Organization; 2024 (https:// iris.who.int/handle/10665/375711). Licence: CC BY-NC-SA 3.0 IGO.
- Immunization dashboard: global [WHO/UNICEF estimates of national immunization coverage (WUENIC), 2024 revision] [website]. Geneva: World Health Organization, United Nations Children's Fund; 2024 (https://immunizationdata.who.int/, accessed on 27 March 2025).
- United Nations Children's Fund (UNICEF), World Health Organization, International Bank for Reconstruction and Development/The World Bank. Levels and trends in child malnutrition: key findings of the 2025 edition of the joint child malnutrition estimates. Geneva: World Health Organization; 2025 [in press]. Licence: CC BY-NC-SA 3.0 IGO.
- 14. WHO global anaemia estimates, 2025 edition. Geneva: World Health Organization; 2025 (https://www.who.int/data/gho/ data/themes/topics/anaemia_in_women_and_children).
- Progress on household drinking water, sanitation and hygiene 2000–2022: special focus on gender. New York: United Nations Children's Fund and World Health Organization; 2023 (https://washdata.org/reports/jmp-2023-wash-households).
- Air pollution data portal. The global health observatory [online database]. Geneva: World Health Organization; 2025 (https://www.who.int/data/gho/data/themes/air-pollution).

Annex 2.

Summary of methodology

Chapter 1

The analysis presented in Chapter 1 is based on the WHO Global health estimates 2021 (GHE2021), which included the latest estimates on life expectancy, healthy life expectancy, causes of deaths and disability globally, by region and country, and by age and sex. The GHE2021 were produced using data from multiple sources, including national vital registration data, latest estimates from WHO technical programmes, United Nations partners and interagency groups, and the Institute for Health Metrics and Evaluation Global Burden of Disease and other scientific studies. The GHE2021 was reviewed by WHO Member States through consultation with national focal points and WHO country and regional offices. The GHE2021 summary of methodology can be accessed online (1).

Secondary analysis was performed using well-established demographic methods to decompose the change and difference in healthy life expectancy over time or between different populations, by age and cause, using GHE2021 data (2,3).

Chapter 2

The analysis presented in Chapter 2 is based on data available from the global monitoring of health-related SDG indicators as of April 2025. Data were compiled from publications and databases produced and managed by WHO or United Nations partner entities. The methodology for each indicator is available from the cited references for the chapter. The data reference years for each indicator differ as data series are updated on different timelines and with different lags between the data reference year and the publication year. Unless otherwise stated, the assessment of progress towards reaching a target is determined by whether the target will be met if the current average annual rate of change applies until the target year.

Chapter 3

The Triple Billion estimates in Chapter 3 were calculated following the WHO GPW13 measurement framework, developed through extensive review and consultation with experts, WHO regional offices and Member States (4). Estimates were derived from 46 outcome indicators (5), using the latest available country-level data as of March 2025. Indicator-level projections were generated using statistical forecasting models that account for historical trends, data sparsity and recent developments, according to methods in Concept and methodologies in estimation and forecasting of Triple Billion targets and improving the WHO results framework (6).

Additionally, two hypothetical scenarios are presented. The first shows estimated progress if all countries were to achieve the annualized targets set by the 2030 Agenda (5,7). This scenario was constructed by calculating a population-weighted global average of expected 2030 values for each indicator. Each country's expected 2030 value was adjusted by the proportion needed to close the gap between the global average and the SDG target/ indicator. The second scenario shows potential gains in progress if countries within each WHO region were to match the expected progress of top countries in their respective regions by 2030. Region-specific targets are defined as the top decile of the regional distribution of 2030 values for each indicator. Country-specific trajectories for the scenario were calculated by finding the annualized rate of change required to meet regional-targets, from a baseline year of 2023.

Chapter 4

Economic status is measured using a wealth index and is composed of five subgroups, each accounting for 20% of the population. Education refers to the highest level of education attained by the mother and is composed of three subgroups (no education, primary education and secondary or higher education). For economic status and education, inequality was measured as the difference along the socioeconomic spectrum, while taking the composition of every socioeconomic subgroup into consideration. This is also known as the slope index of inequality (SII). Subgroups were ranked from the most disadvantaged to the most advantaged, factoring in their population sizes, and the indicator value regressed against these relative ranks. SII represents the absolute difference between the two extremes of this regression line. For place of residence, inequality was measured as the difference between urban and rural subgroups (for DTP3 coverage) and the difference between rural and urban subgroups (for zero-dose prevalence). Since place of residence is categorized into two subgroups (urban and rural), the inequality measured by the difference might be less than when there are more subgroups (as is the case with economic status and education). This is because a larger number of groupings typically captures more heterogeneity. Globally and within World Bank country income groups, inequality was calculated as the median difference across countries. Change in inequality over time was measured using countries with two data points, from 2004 to 2013 and from 2014 to 2023 (spanning approximately 10 years).

References

- 1. Global health estimates: life expectancy and leading causes of death and disability. In: The Global Health Observatory [website]. World Health Organization; (https://www.who.int/data/gho/data/themes/mortality-and-global-health-estimates).
- Andreev EM, Shkolnikov VM, Begun AZ. Algorithm for decomposition of differences between aggregate demographic measures and its application to life expectancies, healthy life expectancies, parity-progression ratios and total fertility rates. Demogr Res. 2002;7:14 (https:// doi.org/10.4054/demres.2002.7.14).
- Beltrán-Sánchez H, Preston SH, Canudas-Romo V. An integrated approach to cause-of-death analysis: causedeleted life tables and decompositions of life expectancy. Demogr Res. 2008;19:1323–1350 (https://doi.org/10.4054/ DemRes.2008.19.35).
- Thirteenth general programme of work (GPW13): methods for impact measurement, version 2.1. Geneva: World Health Organization; 2020 (https://iris.who.int/ handle/10665/341371). Licence: CC BY-NC-SA 3.0 IGO.

- 5. Thirteenth general programme of work (GPW13): metadata for impact measurement indicators. Geneva: World Health Organization; 2020 (https://www.who.int/publications/m/ item/metadata-for-impact-measurement-indicators).
- Concept and methodologies in estimation and forecasting of Triple Billion targets and improving the WHO results framework. Geneva: World Health Organization; 2023 (https://www.who.int/publications/m/item/concept-andmethodologies-in-estimation-and-forecasting-of-triplebillion-targets-and-improving-the-who-results-framework).
- Seventieth session United Nations General Assembly, 25 September 2015. Transforming our world: the 2030 Agenda for Sustainable Development; 2015 (A/RES/70/1; https://docs. un.org/en/A/RES/70/1).



World Health Organization

Avenue Appia 20 1211 Geneva Switzerland http://www.who.int